

INTERNATIONAL JOURNAL OF STRATEGIC PROPERTY MANAGEMENT

2025 Volume 29 Issue 1 Pages 62–80

https://doi.org/10.3846/ijspm.2025.23584

EVALUATING SMART CITY TECHNOLOGY EFFICIENCY AND CITIZEN SATISFACTION USING DATA ENVELOPMENT ANALYSIS

Omer BAFAIL 💿 *

Department of Industrial Engineering, College of Engineering, King Abdulaziz University, Jeddah, Saudi Arabia

Article History:

received 18 November 2024

accepted 21 February 2025

Abstract. This study employs Data Envelopment Analysis (DEA) to evaluate the efficiency of the top 20 smart cities in converting Research and Development (R&D) investments into desired outcomes. Using national R&D expenditure (2015–2022) as input and ten criteria from the IMD 2024 Smart City Index report as outputs, the analysis reveals varying levels of efficiency among leading smart cities. Seven cities achieved perfect efficiency scores, while others, including some high-ranking cities, showed unexpected inefficiencies. This study provides valuable insights into resource utilization and identifies specific areas for improvement across structural and technological dimensions. The limitations include the focus on top-performing cities and the use of national R&D data as a proxy for city-specific investments. The findings of this study offer a foundation for policy-makers and urban planners to optimize resource allocation and improve smart city initiatives, contributing to the ongoing development of sustainable urban environments in the face of technological advancements and urban challenges.

Keywords: smart city efficiency score, human development index, urban planning, citizen satisfaction, sustainable development, assessment method.

*Corresponding author. E-mail: oabafail@kau.edu.sa

1. Introduction

Conceived in the late 20th century, the smart cities concept has continued to evolve as a popular global framework to address challenges posed by rapid urbanization during the 21st century. While its initiatives are increasingly interdisciplinary, the foundation of smart cities rests on successfully implementing innovative uses of Information and Communication Technologies (ICT) to increase efficiency, enhance the quality of life, and promote sustainable growth in cities (Ahvenniemi et al., 2017; Caragliu et al., 2011; Angelidou, 2014; Bibri & Krogstie, 2017; Kramers et al., 2014; Neirotti et al., 2014; Albino et al., 2015; Yigitcanlar et al., 2019; Zanella et al., 2014). The United Nations Department of Economic and Social Affairs estimates that 68% of the world's population will reside in urban areas by 2050 (United Nations, 2019). Lessons learned from the smart city projects are crucial to addressing the complex problems accompanying rapid urban growth (Bibri & Krogstie, 2017) that will require today's cities to transform into smart urban environments (Bibri & Krogstie, 2019).

Smart city initiatives promise to offer innovative solutions to pressing urban issues such as traffic congestion, energy consumption, waste management, and public safety. By incorporating advanced technologies such as the Internet of Things (IoT), Artificial Intelligence (AI), and big data analytics, smart cities strive to create more responsive, resilient, and sustainable urban ecosystems (Allam & Dhunny, 2019). However, implementing smart technologies and initiatives does not guarantee success in achieving optimal outcomes or efficiency in addressing urban challenges (Kitchin, 2015; Vanolo, 2014). Thus, this study investigates a critical aspect of smart city development: the efficiency with which cities convert their resources into desired results. Addressing the problems that urban areas face requires optimal success in their initiatives. However, not every project or initiative a city undertakes can be assured of being a success. Present-day cities that rank high in smart city indices almost certainly are not thriving in every smart venture (Anthopoulos, 2017; Caird & Hallett, 2019; Fernandez-Anez et al., 2018; Huovila et al., 2019). In other words, a city's high ranking does not necessarily indicate that it most efficiently uses its resources to achieve smart city outcomes (Kitchin et al., 2019; Mora et al., 2019; Joss et al., 2017).

Therefore, the success of 21st-century cities in meeting the challenges of continued rapid urbanization rests not only on posing solutions, but also on how efficiently and effectively cities implement their initiatives and programs. In this context, efficiency refers to the ratio of outputs to inputs, which indicates how well a municipality or decision-maker utilizes their resources to achieve desired results. Thus, efficiency is a critical metric as it reflects a

Copyright © 2025 The Author(s). Published by Vilnius Gediminas Technical University

This is an Open Access article distributed under the terms of the Creative Commons Attribution License (http://creativecommons.org/licenses/by/4.0/), which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

city's performance as well as performance relative to the resources at its disposal. High efficiency suggests that a city maximizes the impact of its investments and efforts in smart city initiatives. In contrast, low efficiency demonstrates the need to improve resource allocation or implementation strategies.

The main objective of this study is to gain deeper insights into smart city performance by applying DEA to evaluate the efficiency of resource utilization in smart city initiatives. DEA is a non-parametric method that empirically measures the productive efficiency of Decision-Making Units (DMUs), allows for simultaneous comparisons of multiple input and output variables (Charnes et al., 1978; Emrouznejad & Anouze, 2010; Toloo & Tichý, 2015). In this case, the smart cities are the DMUs within the DEA model, used for comparison of efficiency.

The approach of this study provides a comprehensive view of efficiency that goes beyond simple rankings or individual metrics, offering a nuanced assessment of smart city performance.

By focusing on the often-neglected aspect of efficiency, this study makes a significant contribution to the field of smart city research. The application of DEA to smart city data enables the identification of best practices, facilitates benchmarking among cities, and highlights areas where improvements can enhance overall efficiency. Moreover, by comparing efficiency levels to established smart city indices, this study has the potential to challenge existing assumptions about the relationship between perceived and actual performance in urban development.

To test DEA's utility in assessing smart city efficiency, the following research questions were posed:

- To what extent does the efficiency of converting inputs into desired outputs reflect the efficiency of policymakers in smart city governance and development?
- 2. How can insights derived from DEA be used to improve smart city planning and implementation strategies?

To address the research questions, published data will be extracted and analyzed from peer-reviewed reports focusing on a select group of high-ranking smart cities. The balance of this paper is structured as follows: Section 2 provides a comprehensive literature review, examining existing research on smart city evaluation of efficiency measurement and the application of DEA in urban studies. Section 3 describes the methodology used in this study, including the selection of input and output variables, the data collection process, and the DEA model used. Section 4 presents the findings and analysis of the study, providing insights into the efficiency ratings of various smart cities and identifying patterns and trends within the data. Section 5 discusses the implications of the findings, exploring potential reasons for efficiency disparities, and suggesting strategies for improvement. The final section summarizes the key study findings, acknowledging limitations and proposing directions for future smart city efficiency evaluation research.

2. Relevant literature

The assessment of efficiency in urban development is essential for effective resource allocation and the maximization of benefits for citizens. Smart cities have become increasingly popular with rapid advances in ICT, promising to transform urban landscapes and revolutionize how people live, work, and interact. The chief characteristic of smart city initiatives is the utilization of advanced technologies such as AI, the IoT, and "big data" analytics to improve the quality of life of residents, promote sustainability, and drive economic growth (Kushwah et al., 2024; Pelton & Madry, 2024). Caragliu et al. (2011) made a significant contribution to the field of urban studies by offering a comprehensive definition of smart cities, identifying key characteristics, and analyzing these concepts within the European context. Based on Caragliu et al. (2011) study, there are six main axes that characterize smart cities include smart economy, smart mobility, smart environment, smart people, smart living, and smart governance. The six axes for smart cities offers a comprehensive and structured approach to urban development. By focusing on these key areas, cities can create more livable, efficient, and sustainable environments for their citizens. On other hand, Neirotti et al. (2014), explores the concept of smart cities and their global trends. The research aims to provide a comprehensive understanding of smart cities and investigate their diffusion patterns worldwide such as economic, urban, and geographical variables. Another example, Albino et al. (2015), aimed to clarify the concept of smart cities through an extensive literature review. Albino et al. (2015) developed a classification of smart city application domains, including natural resources, energy, transport, buildings, living, government, and economy. From Albino et al. (2015), an analysis of 70 cities implementing smart city projects and introduce a coverage index to measure the extent of smart initiatives across various domains. The study by Albino et al. (2015) investigates how economic, urban, demographic, and geographical factors influence a city's approach to becoming smarter.

This literature review provides an overview of the current research on smart city evaluation methodologies, explicitly focusing on applying DEA in urban studies and smart city contexts. As the smart city concept continues to evolve, researchers and policymakers are facing the challenge of defining, implementing, and evaluating smart city initiatives. This review explores various approaches to smart city evaluation, including the qualitative and quantitative methodologies developed to assess the performance and impact of smart city projects.

2.1. Current state of smart city evaluation research

The evaluation of smart cities has emerged as a critical area of research due to the substantial investments and high expectations associated with these initiatives worldwide. Numerous researchers have endeavored to define the concept of "smart city" in previous studies. For example, Angelidou (2014), conducted a research to address the lack of consensus on smart city definitions and the limited exploration of strategic planning in this field. Angelidou (2014) recognized the multidisciplinary nature of smart cities and sought to trace their historical development to gain a better understanding of what it means to be "smart" in an urban context. Thus, researchers have proposed various frameworks and methodologies to evaluate smart city performance, impact, and sustainability (Anthopoulos et al., 2019).

A key challenge in smart city evaluation is the lack of a universally accepted definition for a "smart city" (Albino et al., 2015). This ambiguity has resulted in diverse evaluation approaches focusing on different aspects of smart city performance. For example, Giffinger et al. (2007) proposed a framework that assesses smart cities based on six key characteristics: smart economy, smart people, smart governance, smart mobility, smart environment, and smart living. This multidimensional approach has been widely adopted and adapted in subsequent research (Lombardi et al., 2012).

Another significant area of research focuses on developing key performance indicators for smart cities. Huovila et al. (2019) conducted a comparative analysis of standardized indicators for smart sustainable cities, emphasizing the importance of selecting appropriate indicators based on each city's specific context and goals. Similarly, Sharifi (2019) critically reviewed smart city assessment tools, highlighting the need for more comprehensive and context-sensitive evaluation methodologies.

The use of big data and advanced analytics in smart city evaluations has gained prominence in recent years. Kitchin et al. (2015) explored the potential of urban indicators and city benchmarking as tools for smart city governance while cautioning against the overreliance on quantitative metrics at the expense of qualitative assessments. Batty et al. (2012) emphasized integrating diverse data sources and analytical approaches to understand smart city performance better.

Zanella et al. (2014) focuses on the application of IoT technology in urban environments to create smart cities. The researchers explore how IoT can enhance various aspects of city management and improve the quality of life for citizens.

Kitchin et al. (2019) developed a framework that integrates the concepts of citizenship, justice, and rights within the context of smart cities. The authors argue that the right to the smart city extends beyond individual rights to encompass collective or common rights. The study by Kitchin et al. (2019) critically examined the implementation of smart city technologies and their impact on urban life. It raises concerns about privacy, data collection, and the potential for surveillance in smart city environments.

Mora et al. (2019) identified key strategic principles for successful smart city development in Europe. The research employed a multiple case study approach to analyze best practices across various European cities. Researchers have also examined the implications of smart city initiatives, emphasizing the need for evaluation frameworks that consider factors such as privacy, digital inclusion, and citizen participation (Cardullo & Kitchin, 2017; Vanolo, 2014). Hollands (2020) critiqued the technocentric approach to smart city development, advocating for a more citizen-centered evaluation that prioritizes social equity and democratic governance.

The economic impact of smart city initiatives has been another area of focus for evaluation research. Caragliu and Del Bo (2019) analyzed the relationship between smart city policies and urban wealth creation, while Neirotti et al. (2014) examined the factors influencing the development of smart city initiatives across various domains.

Out of all the approaches, one of the most debated smart city assessment frameworks is one proposed by Zhang et al. (2019), which utilized the evaluation index system with references to Maslow's hierarchy of needs. The model aimed to fulfill residents' needs on five levels: physiological, safety, love/belonging, esteem, and selfactualization (Ismagilova et al., 2022; Sharif & Pokharel, 2022; Stübinger & Schneider, 2020). Using the Fuzzy Analytical Hierarchy Process (FAHP) method and questionnaires, the weight of each index is assessed, and 29 cities in China were evaluated (Milošević et al., 2021; Ozkaya & Erdin, 2020; Simonofski et al., 2021). The K-means clustering analysis showed differences in the priorities of smart city construction from the perspective of urban residents.

Similarly, Fang and Shan (2024) further elaborated on a people-centered analysis methodology for smart city assessment. Their model aimed to maximize user experience while selecting the allocation of investments considering various aspects of smart cities (Lai et al., 2020; Neves et al., 2020; Yigitcanlar et al., 2022). The study developed efficiency evaluation and user demand models to test the analysis and determine the smart city's development direction and emphasis (Bellini et al., 2022; Guo & Zhong, 2022; Patrão et al., 2020; Sharifi, 2020). This approach focused more on the dynamic assessment of experienced cities, suggesting that indicators should be changed to consider future public demands and relevant technologies.

Huovila et al. (2019) evaluated and compared various standardized indicators used for assessing smart sustainable cities. The research aimed to provide guidance on which indicators and standards are most appropriate for different contexts and purposes.

Researchers have conducted multiple reviews of existing literature to consolidate and gain a deeper understanding of study findings. These reviews provided additional evaluation frameworks or expanded upon facets of earlier studies. For example, Yigitcanlar et al. (2018) analyzed the existing literature to identify key drivers and outcomes of smart city initiatives. Their review resulted in a multidimensional framework for understanding smart cities by intertwining development drivers with desired outcomes (Yigitcanlar et al., 2018). Similarly, Ahvenniemi et al. (2017) compared sustainable and smart city frameworks and indicators to identify critical differences in urban development approaches. This work highlighted gaps in how each approach addressed various aspects of urban sustainability.

Hodson et al. (2023) prioritized human-centered design by analyzing eight smart city projects to evaluate the social impact of smart city technologies and services. In examining smart cities' methods, challenges, and future directions, the authors' findings revealed gaps in social impact evaluation and proposed criteria for assessing smart cities' technologies and services.

Taking a broader perspective, Caird and Hallett (2019) focused on digital, human, and physical characteristics in assessing smart city progress and called for an evidential approach. They also emphasized the importance of using appropriate, valid, and credible evaluation methodologies (Kashef et al., 2021; Ninčević Pašalić et al., 2021; Tan & Taeihagh, 2020) while admitting that challenges and limitations remain in creating evaluation frameworks.

Angelidou (2015) identified four primary factors influencing smart city development: urban futures, knowledge and an innovation economy, technology push, and application pull. Similarly, Huang and Nazir (2021) focused on evaluating smart cities based on IoT use cases and the challenges posed by rapid urban population growth and increased adoption of IoT devices.

Hodson et al. (2023) examined methods, challenges, and future directions for evaluating the social impact of smart city technologies and services. They analyzed eight smart city projects prioritizing human-centered design across various contexts and development phases. This study identified gaps in social impact evaluation in smart city indices, examined projects, proposed criteria for social impact evaluation in smart cities, and suggested new research directions.

Lytras and Visvizi (2018) examined user adoption patterns in smart city services and suggested the need for sustained interdisciplinary approaches (Lytras & Visvizi, 2018). Lacson et al. (2023) also emphasized the importance of a multidisciplinary approach in future smart city evaluations. The authors identified gaps in city assessments through a scoping review, analysis, and synthesis of existing assessments of developing economies. They proposed incorporating more robust, mixed, and quantitative studies.

A longitudinal bibliometric analysis by Mora et al. (2017) revealed publication trends, influential authors, and papers in the smart city field from 1992 to 2012. In 2017, Bibri and Krogstie (2017) conducted an interdisciplinary literature review of state-of-the-art smart, sustainable cities, identified gaps, and proposed a framework that integrated the studies' strengths. Ruhlandt (2018), emphasized the role of smart cities' governance structures, processes, and challenges. After reviewing existing research, the study proposed an integrated conceptual model to assist in smart city evaluation. Fang and Shan (2024) briefly noted that despite many theoretical and methodological surveys and research, practical applications of smart city evaluations require further exploration.

A significant contribution to the field comes from Zhang et al. (2022), who evaluated new first-tier smart cities in China using the entropy method and Technique for Order Preference by Similarity to an Ideal Solution (TOPSIS). Their study established an evaluation system with five dimensions and 30 indicators, providing insights into the varying performance of smart cities across different aspects such as infrastructure, economy, governance, and environment. This research highlights the uneven development of smart city initiatives and the need for targeted improvements in specific areas.

Another significant contribution to the field comes from Makki and Alqahtani (2024), who identified and analyzed the obstacles that hinder the development of smart cities using Decision Making Trial and Evaluation Laboratory (DEMATEL) method.

Zhang et al. (2022), who evaluated new first-tier smart cities in China using the entropy method and TOPSIS.

The multifaceted nature of smart city development is further emphasized by Toli and Murtagh (2020), who delved into the concept of sustainability within smart city definitions. Their research underscores the importance of considering environmental, social, and economic dimensions in smart city evaluations. Building on this, Chen et al. (2022) focused specifically on social sustainability in smart cities, highlighting the critical role of inclusion, equity, and citizen participation as both inputs and long-term outcomes in urban development.

Kramers et al. (2014) studied the potential of ICTs in assessing energy consumption within urban environments. The study aims to develop an analytical framework that can guide city authorities and ICT companies in assessing, identifying, and implementing effective solutions for sustainable urban development.

Practical implementation challenges in smart city projects are addressed by Mosannenzadeh et al. (2017), who introduced a case-based learning methodology to predict barriers to implementing smart and sustainable urban energy projects. This approach offers valuable insights for policymakers and urban planners in anticipating and overcoming obstacles in smart city initiatives. The technological aspect of smart cities is explored by Bibri (2018), who proposed an analytical framework for leveraging IoT and big data applications to enhance environmental sustainability in urban settings. This research demonstrates the potential of advanced technologies in improving city operations and environmental performance.

Efficient planning and operation of urban logistics and mobility services are critical components of smart city development as they directly impact the accessibility, operational efficiency, and human-centric design of these evolving urban environments. A key factor in this process is the crucial role of Geographic Information System analysis, which has proven invaluable for mapping and analyzing spatial and urban patterns in the context of city and urban transport and logistics planning (Nasser et al., 2021).

Caird and Hallett (2019) developed a comprehensive framework for evaluating smart city initiatives. The researchers recognize the growing importance of smart cities and the need for effective evaluation methods to assess their impact and success. Thus, the authors proposed a structured approach to evaluate smart city projects, considering multiple dimensions such as economic, social, and environmental impacts.

Apostolopoulos and Kasselouris (2022) examined the potential of transport pooling in urban logistics, using the case study of the Thriasio Logistics Centre in Greece. Their research highlighted the importance of innovative solutions to address the challenges of urban freight transport, which is a crucial component of smart city development. Meanwhile, Liu et al. (2023) conducted a comprehensive review of GIS models for sustainable urban mobility planning, identifying the current use, future needs, and potential of these tools.

Bafail (2024) employs random forest and regression analysis, two powerful machine learning techniques, to analyze large datasets collected from various urban systems. These methods are commonly used in smart city research for their ability to handle complex, multidimensional data and provide accurate predictions. The findings from Bafail's study reveal that the human development index is a key predictor of smart city performance. Interesting study by Fernandez-Anez et al. (2018) addresses the complex nature of smart city initiatives and proposed a novel approach to understanding and implementing these strategies.

Other researchers have also used other instrument methods to evaluate industrial cities. For instance, Ye et al. (2022) focused on the methodological aspects of smart city evaluation, including the selection of criteria and the application of multi-criteria decision-making models for ranking smart cities. Kourtzanidis et al. (2021) proposed a novel comprehensive evaluation framework that assesses the impact, performance, and sustainability potential of smart city projects. The framework consists of a set of key performance indicators grouped into five dimensions: impact, efficiency, effectiveness, sustainability, and replicability. The versatility of this framework is demonstrated through its application to three distinct smart city projects, showcasing its ability to provide valuable insights for decision-makers and project managers, regardless of the specific context or focus of the initiatives.

2.2. Application of DEA in urban studies and smart cities

The DEA is a powerful tool for evaluating factors that contribute to the efficiency of urban areas. More recently, DEA has been used to evaluate the efficiency of smart cities (Shen et al., 2022). Initially developed by Charnes et al. (1978), DEA has been widely used in various fields, including urban planning, public service delivery, and environmental management.

In the realm of DEA applications to urban systems, recent studies have expanded the use of this methodology beyond traditional efficiency measurements. For instance, Romão et al. (2018) explored the concept of smart cities as common places for tourists and residents, offering a structural analysis of urban attractiveness determinants. Their work provides valuable insights into how smart city initiatives can enhance the quality of life of both permanent inhabitants and visitors.

A study by Lee et al. (2019b) evaluated the efficiency of transfer stations between bus and subway systems in Seoul using smart card data. The researchers employed DEA to estimate the relative efficiency of transfer stations and conducted Tobit regression analysis to identify factors influencing transfer efficiency. The average efficiency score for 32 major stations was 0.557, with efficiency being proportional to the number of transfer trips and transfer rates.

Another study by Lee et al. (2019a) assessed Transit-Oriented Development (TOD) efficiency in Seoul using the network slacks-based measure DEA model. The study analyzed 352 subway station areas using smartcard and socioeconomic data. The overall efficiency score average was 0.349, with transit design and efficiency scores of 0.453 and 0.245, respectively. The findings highlight the importance of balancing transit design and efficiency for optimal station performance

Worthington and Dollery (2000) applied DEA to assess the efficiency of local government service delivery in Australia, and García-Sánchez (2006) used DEA to analyze the efficiency of street cleaning services in Spanish municipalities. These studies demonstrated the versatility of DEA in handling multiple inputs and outputs, making it well-suited for evaluating complex urban systems.

Lee and Jeong (2023) evaluated the equity of vertical transport system installations in Seoul subway stations for mobility-impaired users. Using DEA and smart card data, the researchers found an average equity score of 0.48 for subway stations. Out of 257 stations, 27 were deemed equitable with a score of 1.0, while the bottom 27 stations averaged 0.19, indicating a need for significant improvement. The study suggests that targeted investments based on station-specific needs can lead to more equitable and efficient improvements.

Lee and Lee (2024) evaluated nightlife attractiveness in Seoul, focusing on Millennials and Generation Z (Gen MZ) all-nighters. Using DEA with smart card data and open data sources, the study analyzed 161 Dong unit areas in Seoul. The average attractiveness score was 0.87, suggesting room for improvement to reach maximum attractiveness. The study found that Gen MZ all-nighters from affluent suburban areas tended to frequent these nightlife hotspots, providing insights into existing and potential future hotspots.

Kourtit et al. (2021) used a highly efficient DEA methodology to establish a safety condition benchmark and rank in 57 global and 14 major European cities (Duan et al., 2020; Fan et al., 2021; Kourtit et al., 2021; Omrani et al., 2020). The analysis revealed a gap in safety data and highlighted the potential effectiveness of implementing efficient and informed safety policies in the cities studied (Manoharan et al., 2023; Van Puyenbroeck et al., 2021). The study demonstrated the possible use of DEA in informing urban planning and policy decisions by identifying the relative efficiency in specific areas. Similarly, Kutty et al. (2022) applied DEA methodology to assess the relative sustainability performance of Europe's top 35 smart cities. Their Double-Frontier Slack-Based Measure DEA model utilized optimistic and pessimistic assumptions, resulting in a more accurate sustainability performance evaluation. The study identified Dublin, Oslo, Zurich, and Amsterdam as the four cities with the highest ratings in the overall sustainability index. The use of DEA in this study highlights its ability to evaluate the interactions between various dimensions of sustainability and smart city development.

Fancello et al. (2014) used DEA to evaluate the efficiency of smart city transportation systems in Italian cities, providing insights into the impact of smart technologies on urban mobility. Smart city DEA application has also gained attention in studies by Zhao et al. (2018) in evaluating the energy efficiency of smart grid systems in Chinese cities and by Mardani et al. (2017) through a comprehensive review of DEA applications in energy and environmental studies.

In contrast, Dyson et al. (2001) pointed out challenges in applying DEA to selecting appropriate input and output variables and interpreting results. Dyson et al. (2001) noted that such caution is especially relevant in smart cities, where careful consideration is required in choosing relevant indicators and interpreting efficiency scores. Despite these challenges, the use of DEA for smart city evaluation continues to evolve and grow. Recent advancements in DEA methodologies, such as network DEA (Dyson et al., 2001) and dynamic DEA (Tone & Tsutsui, 2010), offer new opportunities to capture complex relationships and temporal aspects of smart city systems.

Keles and Alptekin (2023) maintains that DEA offers various benefits in assessing smart cities. Firstly, it provides a comparative efficiency score, enabling cities to be ranked and compared for progress (Mao et al., 2023). Secondly, DEA is flexible regarding the number of inputs and outputs involved, making it possible to evaluate a wide range of factors that define smart city performance. Thirdly, DEA does not limit researchers in terms of the functional form of the production process to estimate the production frontier.

In general, the use of DEA in urban studies and smart cities holds promise for evaluating the efficiency and effectiveness of smart city initiatives. Combining DEA with other analytical techniques such as machine learning and big data analytics promises even more powerful tools for evaluation and decision-making in the future.

The extensive literature review on smart city assessment and the use of DEA in urban studies reveals a dynamic and growing research field. It also highlights significant gaps and opportunities that underscore the importance of this study. While most evaluation systems focus on infrastructure capabilities and service delivery, there is a lack of comprehensive approaches considering citizens' feedback and perceptions about the services provided (Javed et al., 2022; Kirimtat et al., 2020; Singh et al., 2023). In addition, the literature shows that while DEA has been successfully applied in various urban contexts, its application to smart city assessment is still developing. This study directly addresses this concern by focusing on the efficiency of smart city performance, providing valuable insights beyond traditional rankings.

The significance of this study is further highlighted by its potential to inform evidence-based decision-making in urban planning and innovative city development. By identifying efficiency frontiers and best practices, this study can guide policymakers and urban planners in optimizing resource allocation and enhancing the overall efficiency of smart city initiatives. In addition, this study contributes significantly to the literature on smart cities and policy effectiveness by introducing a methodological approach to analyze the IMD 2024 report data. By applying efficiency analysis to the IMD 2024 report dataset, this study offers a fresh perspective on smart city performance that has not been previously explored. Furthermore, this study addresses a critical gap in the literature by establishing a direct link between national R&D investment (as a percentage of GDP) and citizen feedback on smart city indicators, creating a unique input-output relationship. This approach provides a nuanced evaluation of the efficiency of smart city initiatives. Moreover, this study focuses on policymakers' efficiency in converting inputs to desired outputs to yield actionable insights that can inform and enhance decision-making processes. Additionally, the utilization of 40 structure and technology indicators derived from citizen feedback as output measures from the IMD 2024 report dataset adds depth to the assessment of smart city performance, transcending traditional metrics. Consequently, this study not only advances the theoretical understanding of smart city development but also provides practical tools for policymakers to optimize the impact of their investments and strategies in fostering more responsive and efficient urban environments.

3. Methodology

This study utilized the DEA method to evaluate the efficiency of smart cities in converting their resources into the desired outcomes. As a non-parametric linear programming model well-suited for this evaluation, the DEA can handle multiple inputs and outputs simultaneously without specifying their functional relationships (Charnes et al., 1978). The methodology of this study involves eight steps as shown in Figure 1. The study encompasses a sample of 20 cities worldwide, chosen based on their inclusion in reputable smart city rankings and consistent data availability for all selected variables. The data for the analysis primarily come from the IMD Smart City Index 2024 Report (IMD World Competitveness Center, 2024), and are supplemented by other urban databases and official city statistics to ensure comprehensive coverage of input and output factors.



Figure 1. Methodology for the assessment of smart city efficiency using the DEA model

The methodology of this study is a structured eightstep process. Initially, the top 20 cities from the IMD Smart City Index 2024 report were identified as the units for the analysis. The next step involves gathering data on R&D expenditure, specifically the average percentage of gross domestic product (GDP) spent on R&D from 2015-2022 for the countries whose cities are identified as the top 20 ranking in the IMD 2024 report. This was followed by collecting citizen feedback data from the IMD 2024 report, focusing on two main criteria: structure and technology. These criteria are further broken down into sub-criteria, including Health & Safety, Mobility, Activities, Opportunities, and Governance, with the average of these sub-criteria being used for analysis. Each of these sub-criteria is composed of several indicators that provide a more detailed assessment of a smart city's performance in that area. In this study, the average values of these indicators within each sub-criterion were calculated to obtain a comprehensive score for each sub-criterion. These averaged sub-criteria scores were then used as outputs in the DEA model. For instance, the Health & Safety sub-criterion in the structural part includes six indicators such as basic sanitation of the poorest areas, recycling services, public safety, air pollution, medical services provision, and finding housing with rent less or equal than 30% of monthly salary. By averaging these indicator values, a single representative score

for each sub-criterion was obtained, providing a balanced view of a city's performance in that particular aspect.

It is important to note that while multiple outputs were considered in the form of these averaged sub-criteria scores, only one input was used in the model. This input was the average R&D expenditure as a percentage of GDP for countries whose cities were included in the top 20 ranks in the IMD 2024 report. This approach allows for an evaluation of how efficiently cities convert their country's R&D investments into tangible improvements across various aspects of urban life and smart city development.

The study then proceeds to select the DEA outputoriented Variable Returns to Scale (VRS) model for efficiency computation. After calculating the efficiency scores, a benchmarking process conducts to compare the performances of the cities. The results are then thoroughly analyzed, and the findings are interpreted in the context of smart city efficiency. Finally, conclusions are drawn from the analysis, and recommendations are formulated based on the insights gained from the study. This comprehensive methodology allows for a systematic evaluation of how efficiently smart cities utilize their resources to achieve desired outcomes in various aspects of urban life and governance.

3.1. Data collection

The data collection for this study draws upon the comprehensive IMD 2024 Smart City Index report (IMD World Competitveness Center, 2024), a publicly available resource that offers valuable insights into urban development and citizen satisfaction. The IMD report covers various aspects, and this study primarily focuses on the structural and technological dimensions of smart cities. The IMD 2024 report is structured into four key sections: background information, priority areas, attitudes, and response scores from citizens. The fourth section, which forms the core of this analysis, presents response scores across 40 factors, categorized into two main sections: structures and technologies. These sections are further divided into five sub-sections each: health & safety, mobility, activities, opportunities, and governance. Each sub-section contains specific indicators, as detailed in Table 1. These indicators are scored on a scale from 0 to 100, based on comprehensive surveys that capture citizens' feedback on both structural and technological aspects of their urban environment. These data provide a nuanced and citizencentric perspective on smart city performance, enabling a robust evaluation of urban development initiatives and their impact on residents' quality of life through the lens of structural and technological advancements.

As presented in Table 1, the structure pillar measures the quality of a city's infrastructure, while the technology pillar evaluates the availability and use of technological solutions. This index incorporates feedback from 120 residents per city to reflect recent developments.

The key indicators in the IMD 2024 report include digital equity, affordable housing, environmental sustainability,
 Table 1. Input and outputs criteria for smart city DEA model

| Average R&D expenditure as a percentage of GDP (2015–2022) Economic Economic Input 1 Basic sanitation meets the needs of the poorest areas Structure Health & safety Output 1 Recycling services are satisfactory Public safety is not a problem Air pollution is not a problem Mobility Output 2 Public rangerous provision is satisfactory Structure Activates Output 2 Public rangerous is satisfactory Structure Activates Output 3 Caltural actives (shows, bars, and museums) are satisfactory Structure Opportunities Output 4 Mosing percentation on local government projects Opportunities Output 4 Most children have access to a good school Lifelong learning opportunities are provided by local institutions Businesses are creating new jobs Minorities feel welcome Information on local government projects Governance Output 5 Corruption of ority official is to an issue of concem Residents contribute to decision making of local government Health & safety Output 6 A website or app allows residents teg safer A A A Governance Output 7 A public thring portives information on traffic congestion Technology Mob | Indicator | Main criteria | Sub criteria | Feature |
|---|--|---------------|-----------------|-----------|
| Basic sanitation meets the needs of the poorest areasStructureHealth & safetyOutput 1Recycling services are satisfactoryPublic safety is not a problemAir pollution is not a problemMedical services provision is satisfactoryFinding housing with rest 320% of a monthly salary is not a problemStructureMobilityOutput 2Public transport is satisfactoryStructureMobilityOutput 2Public transport is satisfactoryStructureOpportunitiesOutput 3Cultural actives (hows, bars, and museums) are satisfactoryStructureOpportunitiesOutput 4Most children have access to a good schoolLifeton Jearning opportunities are provided by local institutionsStructureGovernanceOutput 5Corruption of city officials is not an issue of concernStructureGovernanceOutput 6Residents contribute to decision making of local governmentResidentsStructureHealth & safetyOutput 6A website or app allows residents to onliced government projectsResidentsHealth & safetyOutput 7CrUC ameras have made residents feel saferYen yen yen yen yen yen yen yen yen yen y | Average R&D expenditure as a percentage of GDP (2015–2022) | Economic | Economic | Input 1 |
| Recycling services are satisfactory Public safety is not a problem Medical services provision is satisfactory Finding housing with rent \$30% of a monthly salary is not a problem Traffic congestion is not a problem Structure Mobility Output 2 Public transport is satisfactory Structure Mobility Output 3 Cultural actives (shows, bars, and museums) are satisfactory Structure Opportunities Output 3 Employment finding services are readily available Structure Opportunities Output 4 Most children have access to a good school Lifelong learning opportunities are provided by local institutions Businesses are cateting new jobs Utput 5 Minorities feel welcome Itructure Governance Output 5 Corruption of local government decision are easily accessible Structure Governance Utput 5 Corruption of city officials is not an issue of concern Residents contribute to decision making of local government Health & safet Output 6 A website or ap allows residents to give away unwanted items easily Free public Wi-Fi has improved access to city services Itructure Valut 7 CYC ameras have made residents feel safer Varany and available parking space | Basic sanitation meets the needs of the poorest areas | Structure | Health & safety | Output 1 |
| Public safety is not a problemAir pollution is not a problem to satisfactoryFinding housing with rent s30% of a monthly salary is not a problemTraffic congestion is not a problemCongent satisfactoryBublic transport is satisfactoryCultural actives (shows, bars, and museums) are satisfactoryEmployment finding services are readily availableMost children have access to a good schoolUffer and we access to a good schoolUffer and the welcomeInformation on local government decision are easily accessibleStructureCorruption of city officials is not an issue of concernResidents contribute to decision making of local governmentResidents contribute to decision making of a pollution effectivelyArranging medical appointments to give away unwanted items easilyCree public Wi-Fi has improved access to thy servicesCTV cameras have made residents feel saferA website or app allows residents to lip wait pollution effectivelyArranging medical appointment sonline has improved accessConstanting app shave reduced congestionOnline scheduling and ticket sales h | Recycling services are satisfactory | | | |
| Air pollution is not a problemVertical services provision is satisfactoryFinding housing with rest 230% of a monthly salary is not a problemStructureMobilityOutput 2Traffic congestion is not a problemStructureMobilityOutput 3Public transport is satisfactoryStructureActivatesOutput 3Cultural actives (shows, bars, and museums) are satisfactoryStructureOpportunitiesOutput 4Most children have access to a good schoolStructureOpportunitiesOutput 5Elfelong learning opportunities are provided by local institutionsStructureGovernanceOutput 5Businesses are creating new jobsStructureGovernanceOutput 5Corruption of city officials is not an issue of concernStructureGovernanceOutput 6Residents contribute to decision making of local governmentTechnologyHealth & safetyOutput 6Online reporting of city maintenance problems provides a speedy solutionTechnologyHealth & safetyOutput 7A website or apa allows residents to give away unwanted items easilyFere public Wi-Fi has improved access to tity servicesFere public Wi-Fi has improved access to city servicesFere public Wi-Fi has improved access to ity servicesFere public Wi-Fi has improved access to item and public on ffectivelyFere public Wi-Fi has improved access to item and public transport more accessible to useFere public Wi-Fi has improved access to item and public transport more accessible to useFere public Wi-Fi has improved access to item and public transport more accessible to useFere public Wi-Fi has improved access t | Public safety is not a problem | | | |
| Medical services provision is satisfactoryVertual structureMobilityOutput 2Finding housing with rent \$20% of a monthly salary is not a problemStructureMobilityOutput 2Varific congestion is not a problemStructureActivatesOutput 3Cultural actives (shows, bars, and museums) are satisfactoryStructureActivatesOutput 4Most children have access to a good schoolStructureOpportunitiesOutput 4Lifelong learning opportunities are provided by local institutionsStructureOpportunitiesOutput 5Businesses are creating new jobsStructureGovernanceOutput 5Corruption of city officials is not an issue of concernStructureGovernanceOutput 5Residents contribute to decision making of local governmentTechnologyHealth & safetyOutput 6A website or app allows residents to give away unwanted items easilyFree public Wi-Fi has improved accessUutput 6A website or app allows residents to monitor air pollution effectivelyKarsenMobilityOutput 7Arrenging medical appointments online has improved accessTechnologyMobilityOutput 6Arrenging medical appointments on traffic congestion intrough mobile phonesTechnologyMobilityOutput 7Online access to job listings has made it easier to thony mobile phonesTechnologyMobilityOutput 6A website or app allows residents to monitor air pollution effectivelyKarsenKarsenKarsenKarsenArrenging medical appointments online has improved access <td>Air pollution is not a problem</td> <td></td> <td></td> <td></td> | Air pollution is not a problem | | | |
| Finding housing with rent \$30% of a monthly salary is not a problemStructureMobilityOutput 2Traffic congestion is not a problemStructureMobilityOutput 3Public transport is satisfactoryStructureActivatesOutput 3Cultural actives (shows, bars, and museums) are satisfactoryStructureOpportunitiesOutput 4Employment finding services are readily availableStructureOpportunitiesOutput 4Most children have access to a good schoolLifelong learning opportunities are provided by local institutionsStructureGovernanceOutput 5Minorities feel welcomeInformation on local government decision are easily accessibleStructureGovernanceOutput 5Corruption of city officials is not an issue of concernResidents provide feedback on local government projectsHealth & safetOutput 6Online reporting of city maintenance problems provides a speedy solutionTechnologyHealth & safetOutput 6A website or app allows residents to give away unwanted items easilyFree public Wi-Fi has improved access to city servicesIterative servicesIterative servicesCTV cameras have made residents feel saferA website or app allows residents to monitor air pollution effectivelyMobilityOutput 8Apps that firet cy ou to an available parking space have reduced journey timeStructureNobilityOutput 8Online excluding and ticket sols have made public transport more accessible to useIterativesOutput 8Online scheduling and ticket sols have made it easier to atat anew businessTe | Medical services provision is satisfactory | | | |
| Traffic congestion is not a problemStructureMobilityOutput 2Public transport is satisfactoryStructureActivatesOutput 3Green spaces are satisfactoryStructureActivatesOutput 3Cultural actives (shows, bars, and museums) are satisfactoryStructureOpportunitiesOutput 4Most children have access to a good schoolStructureOpportunitiesOutput 4Miser spaces are creating new jobsStructureGovernanceOutput 5Minorities feel welcomeInformation on local government decision are easily accessibleStructureGovernanceOutput 5Corruption of city officials is not an issue of concernResidents contribute to decision making of local governmentHealth & safetyOutput 6A website or app allows residents to give away unwanted items easily Free public Wi-Fi has improved access to city servicesHealth & safetyOutput 7Arranging medical appointments online has improved accessTechnologyMobilityOutput 7Apps have reduced congestionTechnologyMobilityOutput 7Apps that direct you to an available parking space have reduced journey timeSince Safet Sa | Finding housing with rent ≤30% of a monthly salary is not a problem | | | |
| Public transport is satisfactoryStructureActivatesOutput 3Green spaces are satisfactoryStructureActivatesOutput 4Cultural actives (shows, bars, and museums) are satisfactoryStructureOpportunitiesOutput 4Most children have access to a good schoolStructureOpportunitiesOutput 5Lifelong learning opportunities are provided by local institutionsStructureGovernanceOutput 5Minorities feel welcomeInformation on local government decision are easily accessibleStructureGovernanceOutput 5Corruption of city officials is not an issue of concernResidents contribute to decision making of local governmentHealth & safetyOutput 5Corruption of city officials is not an issue of concernResidents provide feedback on local government projectsHealth & safetyOutput 5Online reporting of city maintenance problems provides a speedy solutionTechnologyHealth & safetyOutput 6A website or app allows residents to give away unwanted items easilyKespiteKespiteKespiteKespiteCrU cameras have made residents feel saferKespiteKespiteKespiteKespiteKespiteKespiteCar-sharing apps have reduced congestionTechnologyMobilityOutput 7Apps that direct you to an available parking space have reduced journey timeKeivatesKeivatesKeivatesOnline purchasing of tickets to shows and museums has made it easier to atten on the fich onlogyMobilityOutput 7Apps that direct you to an available parking space | Traffic congestion is not a problem | Structure | Mobility | Output 2 |
| Green spaces are satisfactoryStructureActivatesOutput 3Cultural actives (shows, bars, and museums) are satisfactoryStructureOpportunitiesNotEmployment finding services are readily availableStructureOpportunitiesOpportunitiesMost children have access to a good SchoolLifelong learning opportunities are provided by local institutionsStructureOpportunitiesBusinesses are creating new jobsStructureGovernanceOutput 5Minorities feel welcomeInformation on local government decision are easily accessibleStructureGovernanceOutput 5Corruption of city officials is not an issue of concernResidents provide feedback on local government projectsHealth & safetyOutput 6Online reporting of city maintenance problems provides a speedy solutionTechnologyHealth & safetyOutput 7A website or app allows residents to give away unwanted items easilyFree public Wi-Fi has improved access to city servicesKerrerKerrerKerrerCTV cameras have made residents feel saferKerrerKerrerKerrerKerrerKerrerKerrerA website or app allows residents to monitor air pollution effectivelyKerrerKerrerKerrerKerrerKerrerArranging medical appointments online has improved accessFechnologyMobilityOutput 7Apps that direct you to an available parking space have reduced journey timeSite KerrerKerrerKerrerDinine exclusing of tickets to shows and museums has made it easier to attendTechnologyActivates | Public transport is satisfactory | | | |
| Cultural actives (shows, bars, and museums) are satisfactoryStructureOpportunitiesOutput 4Most children have access to a good schoolIf elong learning opportunities are provided by local institutionsStructureOpportunitiesOutput 4Misnestes are creating new jobsMinorities feel welcomeStructureGovernanceOutput 5Corruption on local government decision are easily accessibleStructureGovernanceOutput 5Corruption of city officials is not an issue of concernResidents contribute to decision making of local governmentHealth & safetyOutput 6Residents contribute to decision making of local governmentTechnologyHealth & safetyOutput 6A website or app allows residents to give away unwanted items easilyFree public Wi-Fi has improved access to city servicesFreeFree public Wi-Fi has improved access to city servicesFreeFree public Wi-Fi has improved access to city servicesFree< | Green spaces are satisfactory | Structure | Activates | Output 3 |
| Employment finding services are readily availableStructureOpportunitiesOutput 4Most children have access to a good schoolLifelong learning opportunities are provided by local institutionsStructureNew ComportantiesStructureStructureStructureStructureStructureStructureStructureOutput 5Minorities feel welcomeStructureGovernanceOutput 5StructureGovernanceOutput 5Corruption of city officials is not an issue of concernResidents contribute to decision making of local governmentResidents contribute to decision making of local governmentHealth & safetyOutput 6Residents contribute to decision making of local government projectsOnline reporting of city maintenance problems provides a speedy solutionTechnologyHealth & safetyOutput 6A website or app allows residents feel saferKernanceKernanceKernanceKernanceKernanceArranging medical appointments online has improved accessTechnologyMobilityOutput 7Apps that direct you to an available parking space have reduced journey timeStructureNobilityOutput 8Online scheduling and ticket sales have made public transport more accessible to useFee HonlogyActivatesOutput 8Online purchasing of ticks to shows and museums has made it easier to atten of the subject of bistings has medie te asier to strut a new businessFee HonlogyActivatesOutput 8Online services provided by the city have made it easier to strut a new businessFee HonlogyActivatesOutput 8Online pu | Cultural actives (shows, bars, and museums) are satisfactory | | | |
| Most children have access to a good schoolLifelong learning opportunities are provided by local institutionsBusinesses are creating new jobsMinorities feel welcomeInformation on local government decision are easily accessibleStructureGovernanceOutput 5Corruption of city officials is not an issue of concernResidents contribute to decision making of local governmentHealth & safetyOutput 6Residents contribute to decision making of local governmentHealth & safetyOutput 6A website or app allows residents to give away unwanted items easilyHealth & safetyOutput 6Free public Wi-Fi has improved access to city servicesKerrerKerrerKerrerCCTV cameras have made residents feel saferKerrerKerrerKerrerA website or app allows residents to monitor air pollution effectivelyKerrerKerrerKerrerArraning apps have reduced congestionTechnologyMobilityOutput 7Apps that direct you to an available parking space have reduced journey timeKerrerKerrerKerrerBicycle hiring has reduced congestionTechnologyActivatesOutput 8Online scheduling and ticket sales have made public transport more accessible to useKeritatesKeritatesKeritatesOnline scheduling and ticket sales have made it easier to start a new businessTechnologyActivatesOutput 8Online scheduling and ticket sales have made it easier to start a new businessKeritatesKeritatesKeritatesThe skills are taught well in schoolTechnologyGo | Employment finding services are readily available | Structure | Opportunities | Output 4 |
| Lifelong learning opportunities are provided by local institutions Businesses are creating new jobs Minorities feel welcome Information on local government decision are easily accessible Corruption of city officials is not an issue of concern Residents contribute to decision making of local government Residents reporting of city maintenance problems provides a speedy solution A website or app allows residents to give away unwanted items easily Free public Wi-Fi has improved access to city services CCTV cameras have made residents feel safer A website or app allows residents to monitor air pollution effectively Arranging medical appointments online has improved access Car-sharing apps have reduced congestion Ananging medical appointments online has improved access Correspondent on taffic congestion through mobile phones Online scheduling and ticket sales have made public transport more accessible to use Online scheduling and ticket to shows and museums has made it easier to attend Online access to job listings has made it easier to find work Online services provided by the city have made it easier to start a new business The current internet speed and reliability meet connectivity needs Online public access to city finances has reduced concectivity needs Online voting has increased participation An online platform where residents can propose ideas has improved city life Processing identification documents online has reduced witing times | Most children have access to a good school | | | |
| Businesses are creating new jobs Minorities feel welcome Structure Governance Output 5 Information on local government decision are easily accessible Structure Governance Output 5 Corruption of city officials is not an issue of concern Residents contribute to decision making of local government Hesidents contribute to decision making of local government Hesidents contribute to decision making of local government Hesidents contribute to decision making of local government projects Untput 6 Online reporting of city maintenance problems provides a speedy solution Technology Health & safety Output 6 A website or app allows residents to give away unwanted items easily Free public Wi-Fi has improved access to city services Free public Wi-Fi has improved access to city services Free public Wi-Fi has improved access to city services Free public Wi-Fi has improved access to city services Free public Wi-Fi has improved access to city services Free public Wi-Fi has improved access Free Public Wi-Fi has many enduced congestion Output 7 Apps that direct you to an available parking space have reduced journey time Bicycle hiring has reduced congestion through mobi | Lifelong learning opportunities are provided by local institutions | | | |
| Minorities feel welcomeStructureGovernanceUtiput 5Corruption of city officials is not an issue of concernStructureGovernanceOutput 5Residents contribute to decision making of local governmentResidents contribute to decision making of local governmentHealth & safetVulput 6Residents provide feedback on local government projectsFechnologyHealth & safetVulput 6Online reporting of city maintenance problems provides a speedy solutionTechnologyHealth & safetVulput 6A website or app allows residents to give away unwanted items easilyFree public Wi-Fi has improved access to city servicesFechnologyHealth & safetVulput 6CCTV cameras have made residents feel saferKursanging medical appointments online has improved accessKursanging medical appointments online has improved accessVulput 7Apps that direct you to an available parking space have reduced journey timeBicycle hiring has reduced congestionMobilityOutput 7Online specified information on traffic congestion through mobile phonesTechnologyActivatesOutput 8Online purchasing of tickets to shows and museums has made it easier to atten and businessTechnologyOpportunitiesOutput 9It skills are taught well in schoolTechnologyActivatesOutput 9Output 9Online services provided by the city have made it easier to start a new businessTechnologyOpportunitiesOutput 9It skills are taught well in schoolTechnologyGovernanceVulput 10Online services provided by the city have made it | Businesses are creating new jobs | | | |
| Information on local government decision are easily accessibleStructureGovernanceOutput 5Corruption of city officials is not an issue of concernResidents contribute to decision making of local governmentResidents provide feedback on local government projectsResidents provide feedback on local government projectsNumerical accessibleNumerical ac | Minorities feel welcome | | | |
| Corruption of city officials is not an issue of concernResidents contribute to decision making of local governmentResidents contribute to decision making of local government projectsOnline reporting of city maintenance problems provides a speedy solutionTechnologyA website or app allows residents to give away unwanted items easilyFree public Wi-Fi has improved access to city servicesCCTV cameras have made residents feel saferA website or app allows residents to monitor air pollution effectivelyArranging medical appointments online has improved accessCar-sharing apps have reduced congestionOnline scheduling and ticket sales have made public transport more accessible to useThe city provides information on traffic congestion through mobile phonesOnline access to job listings has made it easier to start a new businessThe sulfils are taught well in schoolOnline services provided by the city have made it easier to start a new businessThe current internet speed and reliability meet connectivity needsOnline voting has increased participationAn enline platform where residents can propose ideas has improved city lifeProcessing identification documents online has reduced waiting times | Information on local government decision are easily accessible | Structure | Governance | Output 5 |
| Residents contribute to decision making of local governmentResidents provide feedback on local government projectsOnline reporting of city maintenance problems provides a speedy solutionTechnologyHealth & safetyOutput 6A website or app allows residents to give away unwanted items easilyFree public Wi-Fi has improved access to city servicesCCTV cameras have made residents feel safer </td <td>Corruption of city officials is not an issue of concern</td> <td></td> <td></td> <td></td> | Corruption of city officials is not an issue of concern | | | |
| Residents provide feedback on local government projectsTechnologyHealth & safetyOutput 6Online reporting of city maintenance problems provides a speedy solutionTechnologyHealth & safetyOutput 6A website or app allows residents to give away unwanted items easilyFree public Wi-Fi has improved access to city servicesFree public Wi-Fi has improved access to monitor air pollution effectivelyFree public Wi-Fi has improved accessFree public Wi-Fi has improved | Residents contribute to decision making of local government | | | |
| Online reporting of city maintenance problems provides a speedy solutionTechnologyHealth & safetyOutput 6A website or app allows residents to give away unwanted items easilyFree public Wi-Fi has improved access to city servicesImage: Control of the service | Residents provide feedback on local government projects | | | |
| A website or app allows residents to give away unwanted items easilyFree public Wi-Fi has improved access to city servicesCCTV cameras have made residents feel saferA website or app allows residents to monitor air pollution effectivelyArranging medical appointments online has improved accessCar-sharing apps have reduced congestionTechnologyMobilityOutput 7Apps that direct you to an available parking space have reduced journey timeBicycle hiring has reduced congestionOnline scheduling and ticket sales have made public transport more accessible to useThe city provides information on traffic congestion through mobile phonesOnline purchasing of tickets to shows and museums has made it easier to attendTechnologyActivatesOutput 8Online access to job listings has made it easier to find workThe current internet speed and reliability meet connectivity needsOnline public access to city finances has reduced corruptionTechnologyGovernanceOutput 10Online voting has increased participationAn online platform where residents can propose ideas has improved city lifeProcessing identification documents online has reduced waiting times | Online reporting of city maintenance problems provides a speedy solution | Technology | Health & safety | Output 6 |
| Free public Wi-Fi has improved access to city servicesImproved accessCCTV cameras have made residents feel saferImproved accessA website or app allows residents to monitor air pollution effectivelyImproved accessArranging medical appointments online has improved accessImproved accessCar-sharing apps have reduced congestionTechnologyMobilityOutput 7Apps that direct you to an available parking space have reduced journey timeImproved accessImproved accessImproved accessOnline scheduling and ticket sales have made public transport more accessible to useImproved access to composition through mobile phonesImproved accessImproved accessOnline purchasing of tickets to shows and museums has made it easier to attendTechnologyActivatesOutput 8Online access to job listings has made it easier to find workTechnologyOpportunitiesOutput 9IT skills are taught well in schoolImproved accessImproved accessImproved accessOnline services provided by the city have made it easier to start a new businessImproved access to city finances has reduced corruptionImproved accessImproved accessOnline public access to city finances has reduced corruptionImproved accessImproved accessImproved accessImproved accessOnline voting has increased participationImproved access to city finances has reduced corruptionImproved accessImproved accessImproved accessOnline public access to city finances has reduced waiting timesImproved accessImproved accessImproved accessImproved access | A website or app allows residents to give away unwanted items easily | | | |
| CCTV cameras have made residents feel saferImage: constraint of the same of t | Free public Wi-Fi has improved access to city services | | | |
| A website or app allows residents to monitor air pollution effectivelyArranging medical appointments online has improved accessCar-sharing apps have reduced congestionTechnologyMobilityOutput 7Apps that direct you to an available parking space have reduced journey timeBicycle hiring has reduced congestionVVVVOnline scheduling and ticket sales have made public transport more accessible to useVVVVVVOnline purchasing of tickets to shows and museums has made it easier to attendTechnologyActivatesOutput 8Output 9VV <td>CCTV cameras have made residents feel safer</td> <td></td> <td></td> <td></td> | CCTV cameras have made residents feel safer | | | |
| Arranging medical appointments online has improved accessTechnologyMobilityOutput 7Apps that direct you to an available parking space have reduced journey timeTechnologyMobilityOutput 7Bicycle hiring has reduced congestionInitial stream of the stream | A website or app allows residents to monitor air pollution effectively | | | |
| Car-sharing apps have reduced congestionTechnologyMobilityOutput 7Apps that direct you to an available parking space have reduced journey time </td <td>Arranging medical appointments online has improved access</td> <td></td> <td></td> <td></td> | Arranging medical appointments online has improved access | | | |
| Apps that direct you to an available parking space have reduced journey timeBicycle hiring has reduced congestionOnline scheduling and ticket sales have made public transport more accessible to useThe city provides information on traffic congestion through mobile phonesOnline purchasing of tickets to shows and museums has made it easier to attendTechnologyActivatesOutput 8Online access to job listings has made it easier to find workTechnologyOpportunitiesOutput 9IT skills are taught well in schoolOutput 9Online services provided by the city have made it easier to start a new businessTechnologyGovernanceOnline public access to city finances has reduced corruptionTechnologyGovernanceOutput 10Online voting has increased participationAn online platform where residents can propose ideas has improved city lifeFechnologyFechnologySovernanceProcessing identification documents online has reduced waiting timesSovernanceSovernanceSovernance | Car-sharing apps have reduced congestion | Technology | Mobility | Output 7 |
| Bicycle hiring has reduced congestionOnline scheduling and ticket sales have made public transport more accessible to use The city provides information on traffic congestion through mobile phonesOnline purchasing of tickets to shows and museums has made it easier to attendTechnologyActivatesOutput 8Online access to job listings has made it easier to find workTechnologyOpportunitiesOutput 9IT skills are taught well in schoolOutput 9TechnologyOpportunitiesOutput 9Online services provided by the city have made it easier to start a new businessTechnologyGovernanceOutput 10Online public access to city finances has reduced corruptionTechnologyGovernanceOutput 10Online voting has increased participationAn online platform where residents can propose ideas has improved city life Processing identification documents online has reduced waiting timesTechnologySovernanceSovernance | Apps that direct you to an available parking space have reduced journey time | | | |
| Online scheduling and ticket sales have made public transport more accessible to useTechnologyActivatesOutput 8Online purchasing of tickets to shows and museums has made it easier to attendTechnologyActivatesOutput 8Online access to job listings has made it easier to find workTechnologyOpportunitiesOutput 9IT skills are taught well in schoolTechnologyOpportunitiesOutput 9Online services provided by the city have made it easier to start a new businessTechnologyGovernanceOutput 10Online public access to city finances has reduced corruptionTechnologyGovernanceOutput 10Online platform where residents can propose ideas has improved city lifeProcessing identification documents online has reduced waiting timesTechnologySovernance | Bicycle hiring has reduced congestion | | | |
| The city provides information on traffic congestion through mobile phonesTechnologyActivatesOutput 8Online purchasing of tickets to shows and museums has made it easier to attendTechnologyActivatesOutput 9Online access to job listings has made it easier to find workTechnologyOpportunitiesOutput 9IT skills are taught well in schoolOnline services provided by the city have made it easier to start a new businessThe current internet speed and reliability meet connectivity needsFechnologyGovernanceOutput 10Online public access to city finances has reduced corruptionTechnologyGovernanceOutput 10Online platform where residents can propose ideas has improved city lifeProcessing identification documents online has reduced waiting timesFechnologyFechnology | Online scheduling and ticket sales have made public transport more accessible to use | | | |
| Online purchasing of tickets to shows and museums has made it easier to attendTechnologyActivatesOutput 8Online access to job listings has made it easier to find workTechnologyOpportunitiesOutput 9IT skills are taught well in schoolOnline services provided by the city have made it easier to start a new businessTechnologyOpportunitiesOutput 9Online public access to city finances has reduced corruptionTechnologyGovernanceOutput 10Online voting has increased participationAn online platform where residents can propose ideas has improved city lifeFechnologyEventEventProcessing identification documents online has reduced waiting timesTimesEventEventEvent | The city provides information on traffic congestion through mobile phones | | | |
| Online access to job listings has made it easier to find workTechnologyOpportunitiesOutput 9IT skills are taught well in schoolOnline services provided by the city have made it easier to start a new businessThe current internet speed and reliability meet connectivity needsFechnologyGovernanceOutput 10Online public access to city finances has reduced corruptionTechnologyGovernanceOutput 10Online voting has increased participationAn online platform where residents can propose ideas has improved city lifeFechnologyFechnologyFechnologyProcessing identification documents online has reduced waiting timesFechnologyFechnologyFechnologyFechnology | Online purchasing of tickets to shows and museums has made it easier to attend | Technology | Activates | Output 8 |
| IT skills are taught well in school Online services provided by the city have made it easier to start a new business The current internet speed and reliability meet connectivity needs Online public access to city finances has reduced corruption Online voting has increased participation An online platform where residents can propose ideas has improved city life Processing identification documents online has reduced waiting times | Online access to job listings has made it easier to find work | Technology | Opportunities | Output 9 |
| Online services provided by the city have made it easier to start a new business The current internet speed and reliability meet connectivity needs Online public access to city finances has reduced corruption Online voting has increased participation An online platform where residents can propose ideas has improved city life Processing identification documents online has reduced waiting times | IT skills are taught well in school | | | |
| The current internet speed and reliability meet connectivity needs Technology Governance Output 10 Online public access to city finances has reduced corruption Technology Governance Output 10 Online voting has increased participation An online platform where residents can propose ideas has improved city life Event State Event State Processing identification documents online has reduced waiting times Event State Event State Event State | Online services provided by the city have made it easier to start a new business | | | |
| Online public access to city finances has reduced corruption Technology Governance Output 10 Online voting has increased participation An online platform where residents can propose ideas has improved city life Processing identification documents online has reduced waiting times Vertical state Vertical state | The current internet speed and reliability meet connectivity needs | | | |
| Online voting has increased participation An online platform where residents can propose ideas has improved city life Processing identification documents online has reduced waiting times | Online public access to city finances has reduced corruption | Technology | Governance | Output 10 |
| An online platform where residents can propose ideas has improved city life Processing identification documents online has reduced waiting times | Online voting has increased participation | | | |
| Processing identification documents online has reduced waiting times | An online platform where residents can propose ideas has improved city life | | | |
| | Processing identification documents online has reduced waiting times | | | |

security, road congestion, and trust in governance. The index also highlights the growing significance of AI in urban planning and management, emphasizing how smart technologies shape various aspects of urban life, from safety and efficiency to inclusivity and overall quality of life.

This study focuses on the top 20 cities ranked in the report, providing a more in-depth analysis of their performance and characteristics. These leading smart cities, which are presented in Table 2. As shown in Table 2, Zurich tops the list with the highest ranking and an impressive HDI score of 0.989, closely followed by Oslo and Canberra, both sharing the second-highest HDI of 0.980. Interestingly, while Singapore ranks 5th overall, it has a lower HDI (0.939) compared to some lower-ranked cities, such as London (8th) and Stockholm (11th), which boast HDI scores of 0.973 and 0.972, respectively. This suggests that factors beyond the HDI influence the overall ranking. At the lower end of the ranking are cities such as Shanghai (19th) and
 Table 2. Population, GNI per capita data for the analyzed cities, and HDI scores with their descriptive statistics

| City | Ranking (DMU) | Population | GNI per capita (PPP \$) in 2022 | HDI score |
|-------------|------------------|------------|--|--------------|
| Zurich | 1 | 410,000 | 69,433 | 0.989 |
| Oslo | 2 | 1,040,000 | 69,190 | 0.980 |
| Canberra | 3 | 400,000 | 49,257 | 0.980 |
| Geneva | 4 | 200,000 | 69,433 | 0.966 |
| Singapore | 5 | 5,940,000 | 88,761 | 0.939 |
| Copenhagen | 6 | 1,350,000 | 62,019 | 0.967 |
| Lausanne | 7 | 140,000 | 69,433 | 0.966 |
| London | 8 | 8,870,000 | 46,624 | 0.973 |
| Helsinki | 9 | 650,000 | 49,522 | 0.960 |
| Abu Dhabi | 10 | 1,480,000 | 74,104 | 0.911 |
| Stockholm | 11 | 950,000 | 56,996 | 0.972 |
| Dubai | 12 | 2,880,000 | 74,104 | 0.911 |
| Beijing | 13 | 20,460,000 | 18,025 | 0.907 |
| Hamburg | 14 | 1,850,000 | 55,340 | 0.972 |
| Prague | 15 | 1,320,000 | 39,945 | 0.960 |
| Taipei City | 16 | 2,720,000 | 44,057 | 0.916 |
| Seoul | 17 | 9,960,000 | 46,026 | 0.952 |
| Amsterdam | 18 | 1,000,000 | 57,278 | 0.962 |
| Shanghai | 19 | 27,060,000 | 18,025 | 0.880 |
| Hong Kong | 20 | 7,550,000 | 62,486 | 0.949 |

Hong Kong (20th), despite Hong Kong having a relatively high HDI of 0.949. The data in Table 2 reveal that a high HDI does not always correlate directly with a higher overall ranking, indicating the complexity of factors involved in urban development and quality of life assessments.

Table 3 presents a comprehensive overview of the economic indicators for countries with cities ranked highly in the IMD 2024 Smart City Index. Table 3 displays the GDP figures for 2015 to 2022, providing a year-by-year breakdown of countries' economic performance. The values in Table 3 were obtained from the International Monetary Fund (2025). Additionally, the table includes the average GDP for this eight-year period, offering insights into each country's overall economic stability and growth. A key feature of the table is the inclusion of average R&D expenditure as a percentage of GDP for the same timeframe. The data for R&D values were collected from the Organization for Economic Co-operation and Development report (OECD, 2025). This metric is particularly relevant as it reflects each nation's commitment to innovation and technological advancement, factors that are crucial in developing smart cities. The data allow for easy comparison between these countries, highlighting their economic trajectories and investment priorities in research and development, which are fundamental to their success in smart city initiatives.

Outputs dataset for this study were meticulously collected from the IMD, ensuring a reliable and authoritative

Table 3. Economic indicators of smart city leaders: GDP and R&D expenditure (2015–2022)

| City | Country | GDP Ava Ava R&D % Ava I | | | | | | | | | | |
|-------------|----------------------|-------------------------|-------|-------|-------|-------|-------|-------|-------|-------|-------------|----------------------------|
| City | country | 2015 | 2016 | 2017 | 2018 | 2019 | 2020 | 2021 | 2022 | GDP | (2015–2022) | expenditure (2015–2022) |
| Zurich | Switzerland | 542 | 558 | 577 | 602 | 633 | 630 | 711 | 786 | 630 | 3.18% | 20 |
| Oslo | Norway | 320 | 327 | 341 | 373 | 379 | 361 | 482 | 532 | 390 | 1.97% | 8 |
| Canberra | Australia | 1129 | 1171 | 1220 | 1290 | 1368 | 1381 | 1557 | 1733 | 1356 | 2% | 24 |
| Geneva | Switzerland | 542 | 558 | 577 | 602 | 633 | 630 | 711 | 786 | 630 | 3.18% | 20 |
| Singapore | Singapore | 482 | 504 | 537 | 586 | 602 | 578 | 719 | 800 | 601 | 2.01% | 12 |
| Copenhagen | Denmark | 292 | 304 | 319 | 332 | 352 | 366 | 412 | 448 | 353 | 2.96% | 10 |
| Lausanne | Switzerland | 542 | 558 | 577 | 602 | 633 | 630 | 711 | 786 | 630 | 3.18% | 20 |
| London | United Kingdom | 2829 | 2911 | 3042 | 3130 | 3336 | 3219 | 3544 | 3980 | 3249 | 2.58% | 84 |
| Helsinki | Finland | 239 | 247 | 260 | 272 | 289 | 295 | 318 | 345 | 283 | 2.85% | 8 |
| Abu Dhabi | United Arab Emirates | 574 | 612 | 628 | 702 | 726 | 614 | 645 | 743 | 656 | 1.22% | 8 |
| Stockholm | Sweden | 483 | 499 | 518 | 541 | 586 | 595 | 657 | 715 | 574 | 3.35% | 19 |
| Dubai | United Arab Emirates | 574 | 612 | 628 | 702 | 726 | 614 | 645 | 743 | 656 | 1.22% | 8 |
| Beijing | China | 17474 | 18849 | 20519 | 22368 | 24404 | 25547 | 28722 | 31678 | 23695 | 2.25% | 533 |
| Hamburg | Germany | 4143 | 4278 | 4473 | 4665 | 4925 | 4880 | 5237 | 5687 | 4786 | 3.05% | 146 |
| Prague | Czech Republic | 376 | 389 | 417 | 443 | 487 | 480 | 525 | 579 | 462 | 1.89% | 9 |
| Taipei City | Taiwan | 1065 | 1098 | 1155 | 1180 | 1238 | 1359 | 1505 | 1654 | 1282 | 3.10% | 40 |
| Seoul | South Korea | 2021 | 2105 | 2217 | 2348 | 2408 | 2482 | 2686 | 2956 | 2403 | 4.53% | 109 |
| Amsterdam | Netherlands | 887 | 918 | 960 | 1014 | 1081 | 1092 | 1215 | 1367 | 1067 | 2.21% | 24 |
| Shanghai | China | 17474 | 18849 | 20519 | 22368 | 24404 | 25547 | 28722 | 31678 | 23695 | 2.25% | 533 |
| Hong Kong | China | 17474 | 18849 | 20519 | 22368 | 24404 | 25547 | 28722 | 31678 | 23695 | 2.25% | 533 |
| | | | | | | | | | | | | |

Note: Figures are in billions of U.S. dollars. Avg. = Average.

| Table 4 | Smart | city | performance | outputs | dataset |
|---------|-------|------|-------------|---------|---------|
|---------|-------|------|-------------|---------|---------|

| City | Ranking | Output 1 | Output 2 | Output 3 | Output 4 | Output 5 |
|---|--|--|--|---|--|---|
| Zurich | 1 | 71 | 68 | 85 | 72 | 72 |
| Oslo | 2 | 60 | 54 | 79 | 72 | 60 |
| Canberra | 3 | 64 | 60 | 81 | 75 | 59 |
| Geneva | 4 | 64 | 56 | 82 | 69 | 66 |
| Singapore | 5 | 71 | 64 | 77 | 74 | 69 |
| Copenhagen | 6 | 61 | 49 | 78 | 72 | 63 |
| Lausanne | 7 | 60 | 49 | 80 | 70 | 62 |
| London | 8 | 43 | 42 | 73 | 62 | 49 |
| Helsinki | 9 | 57 | 59 | 78 | 68 | 58 |
| Abu Dhabi | 10 | 71 | 68 | 85 | 72 | 72 |
| Stockholm | 11 | 52 | 40 | 76 | 65 | 55 |
| Dubai | 12 | 70 | 60 | 84 | 73 | 73 |
| Beijing | 13 | 70 | 50 | 82 | 76 | 67 |
| Hamburg | 14 | 54 | 43 | 73 | 62 | 53 |
| Prague | 15 | 53 | 46 | 64 | 66 | 53 |
| Taipei City | 16 | 66 | 41 | 67 | 66 | 59 |
| Seoul | 17 | 55 | 47 | 61 | 51 | 48 |
| Amsterdam | 18 | 47 | 43 | 68 | 66 | 53 |
| Shanghai | 19 | 73 | 55 | 83 | 79 | 71 |
| Hona Kona | 20 | 49 | 47 | 55 | 61 | 50 |
| Theng heng | - | - | | | - | |
| City | Ranking | Output 6 | Output 7 | Output 8 | Output 9 | Output 10 |
| City Zurich | Ranking 1 | Output 6 78 | Output 7 73 | Output 8 89 | Output 9 80 | Output 10 76 |
| City Zurich Oslo | Ranking 1 2 | Output 6 78 54 | Output 7 73 52 | Output 8 89 81 | Output 9 80 61 | Output 10 76 45 |
| City Zurich Oslo Canberra | Ranking 1 2 3 | Output 6 78 54 58 | Output 7 73 52 39 | Output 8 89 81 78 | Output 9 80 61 60 | Output 10 76 45 43 |
| City Zurich Oslo Canberra Geneva | Ranking 1 2 3 4 | Output 6 78 54 58 59 | Output 7 73 52 39 55 | Output 8 89 81 78 81 | Output 9 80 61 60 64 | Output 10 76 45 43 55 |
| City Zurich Oslo Canberra Geneva Singapore | Ranking 1 2 3 4 5 | Output 6 78 54 58 59 72 | Output 7 73 52 39 55 59 | Output 8 89 81 78 81 81 82 | Output 9 80 61 60 64 75 | Output 10 76 45 43 55 64 |
| City Zurich Oslo Canberra Geneva Singapore Copenhagen | Ranking 1 2 3 4 5 6 | Output 6 78 54 58 59 72 54 | Output 7 73 52 39 55 59 46 | Output 8 89 81 78 81 82 79 | Output 9 80 61 60 64 75 62 | Output 10 76 45 43 55 64 48 |
| City Zurich Oslo Canberra Geneva Singapore Copenhagen Lausanne | Ranking 1 2 3 4 5 6 7 | Output 6 78 54 58 59 72 54 53 | Output 7 73 52 39 55 59 46 51 | Output 8 89 81 78 81 82 79 78 | Output 9 80 61 60 64 75 62 63 | Output 10 76 45 43 55 64 48 50 |
| City Zurich Oslo Canberra Geneva Singapore Copenhagen Lausanne London | Ranking 1 2 3 4 5 6 7 8 | Output 6 78 54 58 59 72 54 53 55 | Output 7 73 52 39 55 59 46 51 57 | Output 8 89 81 78 81 82 79 78 78 75 | Output 9 80 61 60 64 75 62 63 63 65 | Output 10 76 45 43 55 64 48 50 54 |
| City Zurich Oslo Canberra Geneva Singapore Copenhagen Lausanne London Helsinki | Ranking 1 2 3 4 5 6 7 8 9 | Output 6 78 54 58 59 72 54 53 55 55 54 | Output 7 73 52 39 55 59 46 51 57 57 50 | Output 8 89 81 78 81 82 79 78 75 73 | Output 9 80 61 60 64 75 62 63 65 63 | Output 10 76 45 43 55 64 48 50 54 50 54 50 |
| City Zurich Oslo Canberra Geneva Singapore Copenhagen Lausanne London Helsinki Abu Dhabi | Ranking 1 2 3 4 5 6 7 8 9 10 | Output 6 78 54 58 59 72 54 53 55 55 54 78 | Output 7 73 52 39 55 59 46 51 57 50 73 | Output 8 89 81 78 81 82 79 78 75 73 89 | Output 9 80 61 60 64 75 62 63 63 65 63 80 | Output 10 76 45 43 55 64 48 50 54 50 54 50 76 |
| City Zurich Oslo Canberra Geneva Singapore Copenhagen Lausanne London Helsinki Abu Dhabi Stockholm | Ranking 1 2 3 4 5 6 7 8 9 10 11 | Output 6 78 54 58 59 72 54 53 55 55 54 78 52 | Output 7 73 52 39 55 59 46 51 57 50 73 48 | Output 8 89 81 78 81 82 79 78 75 73 89 74 | Output 9 80 61 60 64 75 62 63 63 65 63 80 58 | Output 10 76 45 43 55 64 48 50 54 50 54 50 76 45 |
| City Zurich Oslo Canberra Geneva Singapore Copenhagen Lausanne London Helsinki Abu Dhabi Stockholm Dubai | Ranking 1 2 3 4 5 6 7 8 9 10 11 12 | Output 6 78 54 58 59 72 54 53 55 54 78 52 78 | Output 7 73 52 39 55 59 46 51 57 50 73 48 72 | Output 8 89 81 78 81 82 79 78 79 78 75 73 89 74 90 | Output 9 80 61 60 64 75 62 63 63 65 63 80 58 80 | Output 10 76 45 43 55 64 48 50 54 50 54 50 76 45 73 |
| City Zurich Oslo Canberra Geneva Singapore Copenhagen Lausanne London Helsinki Abu Dhabi Stockholm Dubai Beijing | Ranking 1 2 3 4 5 6 7 8 9 10 11 12 13 | Output 6 78 54 58 59 72 54 53 55 54 78 52 78 79 | Output 7 73 52 39 55 59 46 51 57 50 73 48 72 78 | Output 8 89 81 78 81 82 79 78 79 78 75 73 89 74 90 88 | Output 9 80 61 60 64 75 62 63 65 63 80 58 80 81 | Output 10 76 45 43 55 64 48 50 54 50 54 50 76 45 73 76 |
| City Zurich Oslo Canberra Geneva Singapore Copenhagen Lausanne London Helsinki Abu Dhabi Stockholm Dubai Beijing Hamburg | Ranking 1 2 3 4 5 6 7 8 9 10 11 12 13 14 | Output 6 78 54 58 59 72 54 53 55 54 78 52 78 79 49 | Output 7 73 52 39 55 59 46 51 57 50 73 48 72 78 50 | Output 8 89 81 78 81 82 79 78 75 73 89 74 90 88 77 | Output 9 80 61 60 64 75 62 63 63 65 63 80 58 80 58 80 81 55 | Output 10 76 45 43 55 64 48 50 54 50 54 50 76 45 73 76 45 |
| City Zurich Oslo Canberra Geneva Singapore Copenhagen Lausanne London Helsinki Abu Dhabi Stockholm Dubai Beijing Hamburg Prague | Ranking 1 2 3 4 5 6 7 8 9 10 11 12 13 14 15 | Output 6 78 54 58 59 72 54 53 55 54 78 52 78 78 79 49 57 | Output 7 73 52 39 55 59 46 51 57 50 73 48 72 78 50 50 50 50 | Output 8 89 81 78 81 82 79 78 75 73 89 74 90 88 77 76 | Output 9 80 61 60 64 75 62 63 63 65 63 80 58 80 58 80 81 55 60 | Output 10 76 45 43 55 64 48 50 54 50 54 50 76 45 73 76 45 73 76 45 51 |
| City Zurich Oslo Canberra Geneva Singapore Copenhagen Lausanne London Helsinki Abu Dhabi Stockholm Dubai Beijing Hamburg Prague Taipei City | Ranking 1 2 3 4 5 6 7 8 9 10 11 12 13 14 15 16 | Output 6 78 54 58 59 72 54 53 55 54 78 52 78 79 49 57 71 | Output 7 73 52 39 55 59 46 51 57 50 73 48 72 78 50 50 50 50 50 65 | Output 8 89 81 78 81 82 79 78 75 73 89 74 90 88 74 90 88 77 76 83 | Output 9 80 61 60 64 75 62 63 65 63 80 58 80 81 55 60 67 | Output 10 76 45 43 55 64 48 50 54 50 54 50 76 45 73 76 45 73 76 45 51 67 |
| City Zurich Oslo Canberra Geneva Singapore Copenhagen Lausanne London Helsinki Abu Dhabi Stockholm Dubai Beijing Hamburg Prague Taipei City Seoul | Ranking 1 2 3 4 5 6 7 8 9 10 11 12 13 14 15 16 17 | Output 6 78 54 58 59 72 54 53 55 54 78 78 78 79 49 57 71 | Output 7 73 52 39 55 59 46 51 57 50 73 48 72 78 50 50 50 65 59 | Output 8 89 81 78 81 82 79 78 75 73 89 74 90 88 77 76 83 80 | Output 9 80 61 60 64 75 62 63 65 63 80 58 80 81 55 60 67 59 | Output 10 76 45 43 55 64 48 50 54 50 54 50 76 45 73 76 45 51 67 59 |
| City Zurich Oslo Canberra Geneva Singapore Copenhagen Lausanne London Helsinki Abu Dhabi Stockholm Dubai Beijing Hamburg Prague Taipei City Seoul Amsterdam | Ranking 1 2 3 4 5 6 7 8 9 10 11 12 13 14 15 16 17 18 | Output 6 78 54 58 59 72 54 53 55 54 78 52 78 78 52 78 79 49 57 71 71 53 | Output 7 73 52 39 55 59 46 51 57 50 73 48 72 78 50 50 50 50 50 50 50 50 50 50 50 50 50 | Output 8 89 81 78 81 82 79 78 75 73 89 74 90 88 77 76 83 80 77 | Output 9 80 61 60 64 75 62 63 63 65 63 80 58 80 81 55 60 67 59 61 | Output 10 76 45 43 55 64 48 50 54 50 54 50 76 45 73 76 45 73 76 45 51 67 59 51 |
| City Zurich Oslo Canberra Geneva Singapore Copenhagen Lausanne London Helsinki Abu Dhabi Stockholm Dubai Beijing Hamburg Prague Taipei City Seoul Amsterdam Shanghai | Ranking 1 2 3 4 5 6 7 8 9 10 11 12 13 14 15 16 17 18 19 | Output 6 78 54 58 59 72 54 53 55 54 78 52 78 78 79 49 57 71 71 53 83 | Output 7 73 52 39 55 59 46 51 57 50 73 48 72 78 50 50 50 50 65 50 65 59 54 83 | Output 8 89 81 78 81 82 79 78 75 73 89 74 90 88 77 76 83 80 77 91 | Output 9 80 61 60 64 75 62 63 63 65 63 80 58 80 81 55 60 67 59 61 85 | Output 10 76 45 43 55 64 48 50 54 50 54 50 54 50 76 45 73 76 45 73 76 45 51 67 59 51 81 |

source for the analysis. As previously detailed in Table 1, the study encompasses a comprehensive set of 40 output indicators, each carefully selected to measure various aspects of smart city performance. These indicators provide a multifaceted view of urban development and technological integration across the cities under examination. Table 4 displays the specific outcome data for the cities under study. The average scores for each sub-criterion were also calculated. For example, the first output includes

the average scores for residents' feedback on health and safety for the structural criterion.

3.2. Data envelopment analysis

This study employs the Banker, Charnes, and Cooper (BCC) model (Banker, 1984), which allows variable returns to scale to account for the diverse sizes and contexts of the cities in the sample. This model allows for a more nuanced

assessment of efficiency, considering that smart cities may operate at varying scales of development. DEA has various models. The two most common are the BCC mentioned above and the Charnes-Cooper-Rhodes (CCR). The CCR model assumes Constant Returns to Scale (CRS) (Alidrisi, 2021; Balubaid et al., 2023; Dellnitz & Rödder, 2021; Ebrahimzade Adimi et al., 2021; Kohl & Brunner, 2020), meaning that any input change results in proportional output changes (Amiri et al., 2023; Cooper et al., 2007; Kraidi et al., 2024; Moghaddas et al., 2023; Sarparast et al., 2022; Xiong et al., 2024). For instance, in the case of smart city energy management, if the input is the number of smart matters installed and the output is the energy savings in kilowatt-hours (kWh), then,

- Installing 1,000 smart meters results in 100,000 kWh of energy savings;
- Installing 2,000 smart meters results in 200,000 kWh of energy savings;
- Installing 3,000 smart meters results in 300,000 kWh of energy savings.

The above example indicates that when the number of smart meters increases by two or three times, the energy savings will also increase proportionally. This ratio of input to output remains consistent regardless of the scale of operation. However, complex systems, such as urban environments, do not always exhibit such perfect proportionality. Therefore, the BCC model with VRS is often more suitable for analyzing smart cities.

The VRS indicates that the relationships between inputs and outputs may not be proportional (Mahajan et al., 2024; Tone & Tsutsui, 2010; Zarrin & Brunner, 2023). As the scale of operations changes, the efficiency or productivity may increase, decrease, or remain constant. The concept acknowledges that not all DMUs operate optimally due to imperfect competition, resource constraints, or other external influences. There are three types of VRS:

- Increasing Returns to Scale (IRS): Output increases by a more significant proportion than the input increase;
- Constant Returns to Scale (CRS): Output increases by the same proportion as the input increases;
- Decreasing Returns to Scale (DRS): Output increases by a smaller proportion than the input increase.

To illustrate the differences between the three VRS types, consider a smart city app for citizen engagement. In this scenario, the input is a \$10,000 investment in app development and marketing, and the output is the number of active users in thousands. The VRS scenario is:

- \$10,000 investment results in 5,000 active users (0.5 users per \$);
- \$20,000 investment results in 15,000 active users (0.75 users per \$) – IRS;
- \$30,000 investment results in 20,000 active users (0.67 users per \$) – DRS.

This example shows increasing returns initially as the app gains popularity, followed by decreasing returns as the market becomes saturated. This demonstrates that the relationship between inputs and outputs is not constant in real-world situations, especially in complex systems such as smart cities (Bartolacci et al., 2025; Raith et al., 2022). The BCC model, which considers the VRS, can accurately assess the efficiency and provide a more realistic performance evaluation across different scales of operation. Additionally, input-oriented models in DEA aim to minimize inputs while keeping outputs constant (Moradi et al., 2025; Toloo et al., 2022; Zubir et al., 2024), while output-oriented models in DEA seek maximum outputs while maintaining constant inputs (Alves & Meza, 2023; Liu & Chen, 2022; Moradi et al., 2025). The choice between input and output orientations depends on the DMUs' control over their inputs and outputs (Zubir et al., 2024). The input-oriented model is suitable for cities with more control over its inputs. By comparison, the output-oriented model is applicable if cities have more control over their outputs, such as services and outcomes.

This study used an output-oriented DEA model over an input-oriented approach due to the nature of the variables being considered. The input variable is defined as the average R&D expenditure as a percentage of GDP for the period 2015–2022, specifically for countries whose cities rank in the top 20 of the IMD Smart City Index 2024. On the other hand, the output variables, which include citizen-evaluated service quality metrics, are more susceptible to policy interventions and managerial decisions. This scenario aligns with the fundamental premise of outputoriented DEA models that seek to maximize outputs while keeping inputs constant.

The output-oriented for the BCC model can be viewed in Equation (1) as:

$$Max \sum_{i=1}^{m} u_i \cdot y_{i0} + w \tag{1}$$

subject to:

$$\sum_{j=1}^{n} v_{j} \cdot x_{j0} = 1;$$

$$\sum_{i=1}^{m} u_{i} \cdot y_{ik} - \sum_{j=1}^{n} v_{j} \cdot x_{jk} + w \le 0,$$

where: u_i represents the weight of output *i* for the DMU under analysis; y_{i0} represents the amount of output *i* of the DMU under analysis; v_j represents the weight of input *j* for the DMU under analysis; x_{j0} represents the amount of input *j* of the DMU under analysis; u_i represents the weight of output *i* for the DMU under analysis; y_{ik} represents the amount of output *i* of the other DMUs; x_{jk} represents the amount of input *j* of the other DMUs; *m* represents the number of outputs analyzed; *n* represents the number of inputs analyzed; *w* represents the scalar variable; *k* represents the comparisons across different DMUs.

In the scope of DEA, Overall Technical Efficiency (OTE) and Pure Technical Efficiency (PTE) provide distinct perspectives on a city's performance. OTE, derived from the CCR model, evaluates a city's efficiency without considering its size by comparing it to the best-performing cities regardless of their size. For example, OTE might indicate that a large metropolis with advanced smart infrastructure is 100% efficient in delivering services to citizens, setting a benchmark for other cities to strive towards. In contrast, PTE, calculated from the BCC model, considers the effects of size and assesses a city's efficiency relative to cities of similar size. For instance, PTE might reveal that a mediumsized city, while not as efficient as a large metropolis in absolute terms, operates at 100% efficiency compared to cities of similar scale and resources. This distinction is crucial in analyzing smart cities, as it enables fairer comparisons and helps to identify whether inefficiencies stem from suboptimal scales, as indicated by OTE and PTE, or by managerial and operational factors, as indicated by PTE alone. By examining both OTE and PTE, policymakers can gain a more comprehensive understanding of their cities' performance and identify potential areas for enhancement in the delivery of smart services.

4. Result and analysis

When evaluating the efficiency of smart cities using a DEA model, it is crucial to consider both input and output factors to measure the success of different cities. All indicators are listed in the previous section of this paper in Tables 1–4.

In DEA, different inputs and outputs are often measured in various units (e.g., dollars, hours, and quantities). The use of different measurement units without adjustments can lead to biased or misleading results. Therefore,

Table 5. DEA efficiency scores for smart cities

| City | DMU (Rank) | Efficiency score | Status |
|-------------|------------|---------------------|-------------|
| Zurich | 1 | 1 | Efficient |
| Oslo | 2 | 1 | Efficient |
| Canberra | 3 | 1 | Efficient |
| Geneva | 4 | 0.890 | Inefficient |
| Singapore | 5 | 1 | Efficient |
| Copenhagen | 6 | 0.913 | Inefficient |
| Lausanne | 7 | 0.848 | Efficient |
| London | 8 | 0.599 | Inefficient |
| Helsinki | 9 | 0.770 | Inefficient |
| Abu Dhabi | 10 | 1 | Efficient |
| Stockholm | 11 | 0.697 | Inefficient |
| Dubai | 12 | 1 | Efficient |
| Beijing | 13 | 0.952 | Inefficient |
| Hamburg | 14 | 0.592 | Inefficient |
| Prague | 15 | 0.686 | Inefficient |
| Taipei City | 16 | 0.808 | Inefficient |
| Seoul | 17 | 0.739 | Inefficient |
| Amsterdam | 18 | 0.634 | Inefficient |
| Shanghai | 19 | 1 | Efficient |
| Hong Kong | 20 | 0.479 | Inefficient |

all data points were normalized before performing the DEA efficiency scores.

The DEA-based efficiency scores were used to assess the performance of each city. Table 5 presents the DEA efficiency scores for the top 20 smart cities in the IMD 2024 report. The DEA defines a DMU as efficient when it can produce the maximum output with its available inputs compared to other DMUs in the same group. The efficiency frontier consists of the top-performing DMUs. These DMUs have a perfect efficiency score of 1 (or 100%) and are benchmarks for less efficient DMUs. Less efficient DMUs fall below the efficiency frontier and have an efficiency score of less than 1 (or less than 100%), indicating room for improvement through increased outputs, decreased inputs, or both. Certain cities have demonstrated higher efficiency in converting inputs into valuable outputs, as measured by citizen surveys across multiple factors.

The DEA results reveal that several cities demonstrate optimal efficiency in converting their inputs into desired smart city outcomes. Zurich, Oslo, Canberra, Singapore, Abu Dhabi, Dubai, and Shanghai all achieved an efficiency score of 1, indicating that these cities are operating at the efficiency frontier. These efficient cities span diverse geographical regions, including Europe, Asia, Oceania, and the Middle East, suggesting that smart city excellence is not confined to a particular continent or economic bloc. Notably, these cities have successfully leveraged their resources to maximize their smart city performance, as reflected in their top rankings in the IMD 2024 Smart City Index. Their achievement of perfect efficiency scores underscores their ability to optimally utilize inputs, including R&D expenditures, to produce superior smart city outcomes across various indicators.

Conversely, the analysis identifies several cities that exhibit inefficiencies in their smart city operations. Cities such as Geneva, Copenhagen, London, Helsinki, Stockholm, Beijing, Hamburg, Prague, Taipei City, Seoul, Amsterdam, and Hong Kong demonstrate efficiency scores below 1, indicating suboptimal conversion of inputs to smart city outputs. The degree of inefficiency varies considerably, with efficiency scores ranging from 0.479 (Hong Kong) to 0.952 (Beijing). This variation suggests diverse challenges and opportunities for improvement across these cities. Interestingly, some high-ranking cities in the IMD Smart City Index, such as Geneva (rank 4) and Copenhagen (rank 6), show inefficiencies despite their overall strong performance, highlighting that even top-performing smart cities have room for enhancing their resource utilization. The identification of these inefficiencies provides valuable insights for policymakers and urban planners, pointing to areas where strategic interventions could yield significant improvements in smart city performance.

As indicated earlier, the DEA measures cities' efficiency by comparing them with other cities. Based on the outputoriented model of BCC, outputs are maximized while inputs are kept constant, and for DMUs with low efficiency, the model determines how outputs must be increased in

| City | Output | | | | | | | | | |
|-------------|--------|-------|-------|-------|-------|-------|-------|-------|-------|-------|
| | 1 | 2 | 3 | 4 | 5 | 6 | 7 | 8 | 9 | 10 |
| Zurich | 0 | 0 | 0 | 0 | 0 | 0 | 0 | 0 | 0 | 0 |
| Oslo | 0 | 0 | 0 | 0 | 0 | 0 | 0 | 0 | 0 | 0 |
| Canberra | 0 | 0 | 0 | 0 | 0 | 0 | 0 | 0 | 0 | 0 |
| Geneva | 0.147 | 0.368 | 0 | 0 | 0.140 | 0.495 | 0.38 | 0.381 | 0.486 | 0.520 |
| Singapore | 0 | 0 | 0 | 0 | 0 | 0 | 0 | 0 | 0 | 0 |
| Copenhagen | 0.263 | 0.454 | 0 | 0 | 0.234 | 0.567 | 0.39 | 0.322 | 0.493 | 0.510 |
| Lausanne | 0.236 | 0.408 | 0 | 0 | 0.321 | 0.705 | 0.438 | 0.611 | 0.53 | 0.603 |
| London | 0.960 | 0.937 | 0 | 0.081 | 0.925 | 0.548 | 0.116 | 0.733 | 0.244 | 0.417 |
| Helsinki | 0.303 | 0 | 0 | 0 | 0.465 | 0.626 | 0.436 | 0.914 | 0.490 | 0.593 |
| Abu Dhabi | 0 | 0 | 0 | 0 | 0 | 0 | 0 | 0 | 0 | 0 |
| Stockholm | 0.530 | 0.998 | 0 | 0 | 0.594 | 0.71 | 0.511 | 0.798 | 0.689 | 0.826 |
| Dubai | 0 | 0 | 0 | 0 | 0 | 0 | 0 | 0 | 0 | 0 |
| Beijing | 0.042 | 0.233 | 0 | 0 | 0.133 | 0.045 | 0.022 | 0.055 | 0.045 | 0.051 |
| Hamburg | 0.363 | 0.852 | 0 | 0.066 | 0.656 | 0.835 | 0.389 | 0.537 | 0.833 | 0.784 |
| Prague | 0.418 | 0.427 | 0.484 | 0 | 0.705 | 0.448 | 0.329 | 0.589 | 0.544 | 0.444 |
| Taipei City | 0 | 0.955 | 0.515 | 0.068 | 0.416 | 0.049 | 0.057 | 0.213 | 0.32 | 0.098 |
| Seoul | 0.372 | 0.372 | 0.678 | 0.835 | 0.984 | 0 | 0.201 | 0.441 | 0.655 | 0.255 |
| Amsterdam | 0.734 | 0.661 | 0.127 | 0 | 0.550 | 0.544 | 0 | 0.314 | 0.406 | 0.295 |
| Shanghai | 0 | 0 | 0 | 0 | 0 | 0 | 0 | 0 | 0 | 0 |
| Hong Kong | 0.599 | 0.017 | 0.924 | 0.231 | 0.728 | 0.310 | 0.238 | 0.616 | 0 | 0.181 |

Table 6. Output slack values in smart city DEA analysis: potential for additional performance improvements

order to reach the efficiency frontier. An inefficient DEA result indicates that the city is not producing the maximum amount of outputs given its inputs in comparison with other cities in the analysis group. For example, Geneva, has an efficiency score of 0.890, which is equivalent to 89%, meaning that it needs to increase its output by 12.3% (1/0.890 = 1.123) to be efficient. Another example, Copenhagen has an efficiency score of 0.913, which indicates it needs to increase outputs by 9.5% (1/0.913 = 1.095) while maintaining its inputs constant in order to achieve efficiency. However, this does not mean Geneva or Copenhagen perform poorly overall but this suggests that there is room for improvement in terms of efficiency compared with the most efficient cities.

While Geneva and Copenhagen may excel in many aspects of being smart cities, the DEA results indicate the possibility of enhancing resource utilization and maximizing output. This insight applies not only to Geneva or Copenhagen but also to all cities found to be inefficient.

Further insights into these inefficiencies can be gleaned through slack analysis in DEA, which provides a more granular view of potential enhancements beyond basic efficiency scores. Slack analysis identifies both input and output slacks, with output slacks being particularly relevant in the context of smart cities as they quantify the potential for increased outputs while maintaining constant inputs. Table 6, which presents output slack values, offers valuable information for policymakers and urban planners by quantifying the additional improvements required in specific smart city outcomes. High output slack values indicate significant potential for enhancement in particular areas, providing a roadmap for targeted interventions to optimize smart city performance across all dimensions, even for top-ranked cities such as Geneva and Copenhagen.

5. Discussion

The DEA of smart cities, using R&D expenditure as a percentage of GDP (2015–2022) as input, reveals significant insights into the efficiency of various urban centers in converting R&D investments into tangible smart city outcomes. The output slack values presented in Table 6 offer a nuanced view of potential improvements across ten critical smart city criteria. This analysis provides valuable insights for policymakers and urban planners, highlighting areas in which strategic interventions could yield substantial benefits.

Structure Health & Safety criterion (Output 1): the analysis reveals varying levels of efficiency across cities in translating R&D investments into structural health and safety improvements. Cities such as Zurich, Oslo, and Canberra demonstrate optimal efficiency with zero slack, indicating their success in this domain. However, cities such as London (0.960) and Amsterdam (0.734) showed significant room for improvement. This suggests that these cities could potentially enhance their health and safety infrastructure without additional R&D investment, perhaps by adopting best practices from more efficient counterparts or by reallocating resources more effectively within this sector.

Structure Mobility criterion (Output 2): in the realm of structural mobility, several cities exhibit substantial potential for improvement. Stockholm (0.998) and London (0.937) showed the highest slack values, indicating significant opportunities to enhance their mobility infrastructure. This could involve improving public transportation systems, implementing smart traffic management solutions, or developing more efficient urban planning strategies. Cities with zero slack, such as Zurich and Singapore, could serve as benchmarks for best practices in this area.

Structure Activities criterion (Output 3): the slack values for this criterion are notably lower across most cities, with many showing zero slack. However, cities like Seoul (0.678) and Taipei City (0.515) demonstrate room for improvement in providing structural support for urban activities. This might involve enhancing public spaces, cultural facilities, or recreational areas to better serve the population's needs and improve the overall quality of life.

Structure Opportunities criterion (Output 4): most cities show minimal slack in this area, suggesting generally efficient conversion of R&D investments into structural opportunities. However, Seoul (0.835) stands out with a high slack value, indicating the potential for significant improvement in creating structural foundations for economic and social opportunities. This could involve developing better educational facilities, business incubators, or job training centers.

Structure Governance criterion (Output 5): the analysis reveals varied performance in governance structures. Cities like Helsinki (0.465) and Stockholm (0.594) show moderate room for improvement, suggesting potential enhancements in e-governance systems, citizen engagement platforms, or administrative efficiency. Cities with zero slack, such as Zurich and Singapore, could offer valuable insights into effective governance structures that maximize R&D investments.

Technology Health & Safety criterion (Output 6): this area shows significant variation across cities. Lausanne (0.705) and Stockholm (0.71) exhibit the highest slack values, indicating substantial potential for technological improvements in health and safety. This could involve the implementation of advanced emergency response systems, health monitoring technologies, or smart surveillance for public safety. Cities such as Zurich and Dubai, with zero slack, may serve as exemplars in this domain.

Technology Mobility criterion (Output 7): the slack values for technological mobility solutions are generally lower, suggesting relatively efficient R&D utilization in this area. However, cities like Stockholm (0.511) and Lausanne (0.438) show room for improvement. This could involve enhancing smart traffic systems, developing more sophisticated public transport applications, or implementing advanced ride-sharing platforms to improve urban mobility.

Technology Activities criterion (Output 8): several cities show significant potential for improvement in this area, with Helsinki (0.914) and Stockholm (0.798) having the highest slack values. This suggests opportunities to better leverage technology to enhance urban activities, possibly through improved digital platforms for cultural events, smart tourism solutions, or advanced systems for managing public spaces and recreational facilities.

Technology Opportunities criterion (Output 9): the analysis reveals varied performance in creating technological opportunities. Cities like Hamburg (0.833) and Stockholm (0.689) show substantial room for improvement. This could involve developing more advanced digital skills training programs, creating better online platforms for job matching, or implementing more sophisticated systems to support entrepreneurship and innovation.

Technology Governance criterion (Output 10): in the realm of technological governance solutions, several cities demonstrate significant potential for enhancement. Stockholm (0.826) and Hamburg (0.784) show the highest slack values, indicating opportunities to improve e-governance platforms, digital citizen engagement tools, or data-driven decision-making systems in urban management.

In conclusion, this DEA analysis provides a comprehensive overview of how effectively cities convert R&D investments into smart city outcomes across various structural and technological domains. While some cities, such as Zurich and Singapore, consistently demonstrate high efficiency, others show varying degrees of potential for improvement. These findings underscore the importance of not only increasing R&D investments but also ensuring their efficient utilization across all aspects of smart city development. Policymakers and urban planners should focus on learning from high-performance cities and adapting successful strategies to their local contexts. Furthermore, cities with high slack values in specific areas should prioritize these domains for targeted interventions and resource reallocation to maximize the impact of their R&D investments on smart city outcomes.

6. Conclusions

This study employed DEA to evaluate the efficiency of the top 20 smart cities in converting their resources into desired outcomes. The input variable was the average R&D expenditure as a percentage of GDP for the period 2015–2022, while the outputs were derived from the IMD 2024 Smart City Index report, encompassing ten criteria across structure and technology dimensions.

The analysis revealed that seven cities – Zurich, Oslo, Canberra, Singapore, Abu Dhabi, Dubai, and Shanghai – achieved perfect efficiency scores, demonstrating optimal utilization of their R&D investments in producing smart city outcomes. Surprisingly, some high-ranking cities in the overall IMD Smart City Index, such as Geneva (4th) and Copenhagen (6th), showed inefficiencies despite their strong performance, highlighting the complexity of smart city development. Data for this study were collected from reputable sources, including national statistics for R&D expenditure and the IMD Smart City Index 2024 for output measures. The DEA methodology provides valuable insights into the relative efficiencies of these cities and identifies specific areas for potential improvement through slack analysis.

However, this study has several limitations that should be acknowledged. Firstly, the focus on the top 20 smart cities limits the generalizability of findings to a broader range of urban environments. Secondly, the use of national R&D expenditure as an input may not fully capture city-specific investments in smart initiatives. Thus, one of the primary limitations is the difficulty in obtaining reliable and consistent data on city-specific investments in R&D for smart initiatives. Many cities do not have uniform methods for reporting their R&D expenditures specifically related to smart city initiatives. Detailed city-level data on R&D investments in smart technologies are often not publicly accessible or may be incomplete. Due to these constraints in acquiring reliable city-specific data, this study utilized national R&D expenditure as a percentage of GDP as a proxy input variable. While this approach provides a broader context for a country's commitment to innovation and technological advancement, it is important to acknowledge that it may not fully capture the nuanced investments made specifically for smart city initiatives at the municipal level.

It is important to note that this study is specifically based on data from the IMD Smart City Index 2024. This temporal specificity is a crucial aspect of the research and has implications for the interpretation and application of the results. The smart city landscape is dynamic and rapidly evolving, with cities constantly implementing new initiatives and technologies. Consequently, the efficiency scores and rankings derived from this analysis are a snapshot of the performance at a particular point in time.

Moreover, the use of data from a single year (2024) implies that the results are sensitive to the specific conditions, achievements, and challenges faced by cities during that period. Any changes in the underlying data, whether due to updated measurements, revised methodologies, or actual changes in city performance, could potentially alter the efficiency scores and rankings.

Future studies could benefit from a longitudinal approach, analyzing data over multiple years to identify trends and the impact of sustained smart city investments. This would provide a more comprehensive understanding of how cities' efficiencies evolve over time and how they respond to various interventions and global challenges.

Funding

This work received no external funding.

Disclosure statement

The author declares no conflicts of interest.

References

- Ahvenniemi, H., Huovila, A., Pinto-Seppä, I., & Airaksinen, M. (2017). What are the differences between sustainable and smart cities? *Cities*, 60, 234–245. https://doi.org/10.1016/j.cities.2016.09.009
- Albino, V., Berardi, U., & Dangelico, R. M. (2015). Smart cities: Definitions, dimensions, performance, and initiatives. *Journal* of Urban Technology, 22(1), 3–21.
 - https://doi.org/10.1080/10630732.2014.942092
- Alidrisi, H. (2021). The development of an efficiency-based global green manufacturing innovation index: An input-oriented DEA approach. *Sustainability*, *13*(22), Article 12697. https://doi.org/10.3390/su132212697
- Allam, Z., & Dhunny, Z. A. (2019). On big data, artificial intelligence and smart cities. *Cities*, *89*, 80–91. https://doi.org/10.1016/j.cities.2019.01.032
- Alves, C. G. M. de F., & Meza, L. A. (2023). A review of network DEA models based on slacks-based measure: Evolution of literature, applications, and further research direction. *International Transactions in Operational Research*, 30(6), 2729–2760. https://doi.org/10.1111/itor.13284
- Amiri, M., Rostamy-Malkhalifeh, M., Lotfi, F. H., & Mozaffari, M. (2023). Measuring returns to scale based on the triangular fuzzy DEA approach with different views of experts: Case study of Iranian insurance companies. *Decision Making: Applications in Management and Engineering*, 6(2), 787–807. https://doi.org/10.31181/dmame622023740
- Angelidou, M. (2014). Smart city policies: A spatial approach. Cities, 41, S3–S11. https://doi.org/10.1016/j.cities.2014.06.007
- Angelidou, M. (2015). Smart cities: A conjuncture of four forces. *Cities*, 47, 95–106. https://doi.org/10.1016/j.cities.2015.05.004
- Anthopoulos, L. G. (2017). Understanding smart cities: A tool for smart government or an industrial trick? (Vol. 22). Springer. https://doi.org/10.1007/978-3-319-57015-0
- Anthopoulos, L., Janssen, M., & Weerakkody, V. (2019). A Unified Smart City Model (USCM) for smart city conceptualization and benchmarking. Smart cities and smart spaces: Concepts, methodologies, tools, and applications (pp. 247–264). IGI Global Scientific Publishing. https://doi.org/10.4018/978-1-5225-7030-1.ch011
- Apostolopoulos, V., & Kasselouris, G. (2022). Seizing the potential of transport pooling in urban logistics the case of Thriasio logistics centre in Greece. *Journal of Applied Research on Industrial Engineering*, 9(2), 230–248.

https://doi.org/10.22105/jarie.2021.309116.1390

Bafail, O. (2024). Optimizing smart city strategies: A data-driven analysis using random forest and regression analysis. *Applied Sciences*, *14*(23), Article 11022.

https://doi.org/10.3390/app142311022

- Balubaid, M., Gulzar, W. A., Aburas, H., Taylan, O., Alkabaa, A. S., Bafail, O. A., Makki, A. A., Alqahtani, A. Y., Alidrisi, H. M., Alsasi, B. O., Karuvatt, S. A., & Alidrisi, H. (2023). Monitoring the performance of agricultural and food secton companies using DEA. *International Journal of Ecosystems and Ecology Science*, 13(2), 9–24. https://doi.org/10.31407/ijees13.2
- Banker, R. D. (1984). Estimating most productive scale size using data envelopment analysis. *European Journal of Operational Research*, *17*(1), 35–44.

https://doi.org/10.1016/0377-2217(84)90006-7

Bartolacci, F., Del Gobbo, R., & Soverchia, M. (2025). Improving public services' performance measurement systems: Applying data envelopment analysis in the big and open data context. *International Journal of Public Sector Management*, 38(3), 313– 331. https://doi.org/10.1108/IJPSM-06-2023-0186 Batty, M., Axhausen, K. W., Giannotti, F., Pozdnoukhov, A., Bazzani, A., Wachowicz, M., Ouzounis, G., & Portugali, Y. (2012). Smart cities of the future. *The European Physical Journal Special Topics*, *214*, 481–518.

https://doi.org/10.1140/epjst/e2012-01703-3

- Bellini, P., Nesi, P., & Pantaleo, G. (2022). IoT-enabled smart cities: A review of concepts, frameworks and key technologies. *Applied Sciences*, 12(3), Article 1607. https://doi.org/10.3390/app12031607
- Bibri, S. E. (2018). The IoT for smart sustainable cities of the future: An analytical framework for sensor-based big data applications for environmental sustainability. *Sustainable Cities and Society*, 38, 230–253. https://doi.org/10.1016/j.scs.2017.12.034
- Bibri, S. E., & Krogstie, J. (2017). Smart sustainable cities of the future: An extensive interdisciplinary literature review. *Sustainable Cities and Society*, *31*, 183–212. https://doi.org/10.1016/j.scs.2017.02.016

Bibri, S. E., & Krogstie, J. (2019). Generating a vision for smart sustainable cities of the future. A scholarly backcasting approach

- tainable cities of the future: A scholarly backcasting approach. *European Journal of Futures Research*, 7(1), Article 5. https://doi.org/10.1186/s40309-019-0157-0
- Caird, S. P., & Hallett, S. H. (2019). Towards evaluation design for smart city development. *Journal of Urban Design*, 24(2), 188– 209. https://doi.org/10.1080/13574809.2018.1469402
- Caragliu, A., & Del Bo, C. (2019). Smart innovative cities: The impact of smart city policies on urban innovation. *Technological Forecasting and Social Change*, 142, 373–383. https://doi.org/10.1016/j.techfore.2018.07.022
- Caragliu, A., Del Bo, C., & Nijkamp, P. (2011). Smart cities in Europe. Journal of Urban Technology, 18(2), 65–82. https://doi.org/10.1080/10630732.2011.601117
- Cardullo, P., & Kitchin, R. (2017). Being a 'citizen'in the smart city: Up and down the scaffold of smart citizen participation (The Programmable City Working Paper 30). https://doi.org/10.31235/osf.io/v24jn
- Charnes, A., Cooper, W. W., & Rhodes, E. (1978). Measuring the efficiency of decision making units. *European Journal of Operational Research*, 2(6), 429–444. https://doi.org/10.1016/0377-2217(78)90138-8
- Chen, T., Ramon Gil-Garcia, J., & Gasco-Hernandez, M. (2022). Understanding social sustainability for smart cities: The importance of inclusion, equity, and citizen participation as both inputs and long-term outcomes. *Journal of Smart Cities and Society*, *1*, 135–148. https://doi.org/10.3233/SCS-210123
- Cooper, W. W., Seiford, L. M., & Tone, K. (2007). Alternative DEA models. In W. W. Cooper, L. M. Seiford, & K. Tone (Eds.), Data envelopment analysis: A comprehensive text with models, applications, references and DEA-Solver software (pp. 87–130). Springer. https://doi.org/10.1007/978-0-387-45283-8_4
- Dellnitz, A., & Rödder, W. (2021). Returns to scale as an established scaling indicator: Always a good advisor? Jahrbücher für Nationalökonomie und Statistik, 241(2), 173–186. https://doi.org/10.1515/jbnst-2019-0058
- Duan, Y.-Q., Fan, X.-Y., Liu, J.-C., & Hou, Q.-H. (2020). Operating efficiency-based data mining on intensive land use in smart city. *IEEE Access*, *8*, 17253–17262.

https://doi.org/10.1109/ACCESS.2020.2967437

- Dyson, R. G., Allen, R., Camanho, A. S., Podinovski, V. V, Sarrico, C. S., & Shale, E. A. (2001). Pitfalls and protocols in DEA. *European Journal of Operational Research*, *132*(2), 245–259. https://doi.org/10.1016/S0377-2217(00)00149-1
- Ebrahimzade Adimi, M., Rostamy-Malkhalifeh, M., Hosseinzadeh Lotfi, F., & Mehrjoo, R. (2021). A model to evaluate the effects of the returns to scale on the inverse data envelopment analy-

sis. Mathematical Sciences, 15(2), 111–121.

- https://doi.org/10.1007/s40096-020-00353-6
- Emrouznejad, A., & Anouze, A. L. (2010). Data envelopment analysis with classification and regression tree–a case of banking efficiency. *Expert Systems*, *27*(4), 231–246.
 - https://doi.org/10.1111/j.1468-0394.2010.00516.x
- Fan, S., Peng, S., & Liu, X. (2021). Can smart city policy facilitate the low-carbon economy in China? A quasi-natural experiment based on pilot city. *Complexity*, 2021(1), Article 9963404. https://doi.org/10.1155/2021/9963404
- Fancello, G., Uccheddu, B., & Fadda, P. (2014). Data Envelopment Analysis (D.E.A.) for urban road system performance assessment. *Procedia - Social and Behavioral Sciences*, *111*, 780–789. https://doi.org/10.1016/j.sbspro.2014.01.112
- Fang, Y., & Shan, Z. (2024). Optimising smart city evaluation: A people-oriented analysis method. *IET Smart Cities*, 6(1), 41–53. https://doi.org/10.1049/smc2.12073
- Fernandez-Anez, V., Fernández-Güell, J. M., & Giffinger, R. (2018). Smart city implementation and discourses: An integrated conceptual model. The case of Vienna. *Cities*, 78, 4–16. https://doi.org/10.1016/j.cities.2017.12.004
- García-Sánchez, I. M. (2006). Efficiency measurement in Spanish local government: The case of municipal water services. *Review of Policy Research*, *23*(2), 355–372.

https://doi.org/10.1111/j.1541-1338.2006.00205.x

- Giffinger, R., Fertner, C., Kramar, H., Kalasek, R., Pichler-Milanovic, N., & Meijers, E. J. (2007). *Smart cities: Ranking of European medium-sized cities* (Final Report). Centre of Regional Science.
- Guo, Q., & Zhong, J. (2022). The effect of urban innovation performance of smart city construction policies: Evaluate by using a multiple period difference-in-differences model. *Technological Forecasting and Social Change*, 184, Article 122003. https://doi.org/10.1016/j.techfore.2022.122003
- Hodson, E., Vainio, T., Sayún, M. N., Tomitsch, M., Jones, A., Jalonen, M., Börütecene, A., Hasan, M. T., Paraschivoiu, I., Wolff, A., Yavo-Ayalon, S., Yli-Kauhaluoma, S., & Young, G. W. (2023). Evaluating social impact of smart city technologies and services: Methods, challenges, future directions. *Multimodal Technologies and Interaction*, 7(3), Article 33. https://doi.org/10.3390/mti7030033
- Hollands, R. G. (2020). Will the real smart city please stand up?: Intelligent, progressive or entrepreneurial? In *The Routledge companion to smart cities* (pp. 179–199). Routledge. https://doi.org/10.4324/9781315178387-13
- Huang, C., & Nazir, S. (2021). Analyzing and evaluating smart cities for IoT based on use cases using the analytic network process. *Mobile Information Systems*, 2021(1), Article 6674479. https://doi.org/10.1155/2021/6674479
- Huovila, A., Bosch, P., & Airaksinen, M. (2019). Comparative analysis of standardized indicators for Smart sustainable cities: What indicators and standards to use and when? *Cities*, *89*, 141–153. https://doi.org/10.1016/j.cities.2019.01.029
- IMD World Competitveness Center. (2024). *IMD Smart City Index* 2024. https://issuu.com/docs/e7a60c053affbf9e98fcba93afe85 7af?fr=xKAE9_zU1NQ
- International Monetary Fund. (2025). GDP, current prices. https:// www.imf.org/external/datamapper/PPPGDP@WEO/OEMDC/ ADVEC/WEOWORLD
- Ismagilova, E., Hughes, L., Rana, N. P., & Dwivedi, Y. K. (2022). Security, privacy and risks within smart cities: Literature review and development of a smart city interaction framework. *Information Systems Frontiers*, 24, 393–414. https://doi.org/10.1007/s10796-020-10044-1

Javed, A. R., Shahzad, F., ur Rehman, S., Zikria, Y. B., Razzak, I., Jalil, Z., & Xu, G. (2022). Future smart cities: Requirements, emerging technologies, applications, challenges, and future aspects. *Cities*, *129*, Article 103794.

https://doi.org/10.1016/j.cities.2022.103794

- Joss, S., Cook, M., & Dayot, Y. (2017). Smart cities: Towards a new citizenship regime? A discourse analysis of the British Smart City Standard. *Journal of Urban Technology*, 24(4), 29–49. https://doi.org/10.1080/10630732.2017.1336027
- Kashef, M., Visvizi, A., & Troisi, O. (2021). Smart city as a smart service system: Human-computer interaction and smart city surveillance systems. *Computers in Human Behavior*, 124, Article 106923. https://doi.org/10.1016/j.chb.2021.106923
- Keles, E. U., & Alptekin, G. I. (2023). Evaluation of the digitalization efficiency of countries using data envelopment analysis. In 2023 Smart City Symposium Prague (SCSP) (pp. 1–5). IEEE. https://doi.org/10.1109/SCSP58044.2023.10146126
- Kirimtat, A., Krejcar, O., Kertesz, A., & Tasgetiren, M. F. (2020). Future trends and current state of smart city concepts: A survey. *IEEE Access*, 8, 86448–86467.

https://doi.org/10.1109/ACCESS.2020.2992441

- Kitchin, R. (2015). Making sense of smart cities: Addressing present shortcomings. *Cambridge Journal of Regions, Economy and Society*, 8(1), 131–136. https://doi.org/10.1093/cjres/rsu027
- Kitchin, R., Cardullo, P., & Di Feliciantonio, C. (2019). Citizenship, justice, and the right to the smart city. In P. Cardullo, C. Di Feliciantonio, & R. Kitchin (Eds.), *The right to the smart city* (pp. 1–24). Emerald Publishing Limited. https://doi.org/10.1108/978-1-78769-139-120191001
- Kitchin, R., Lauriault, T. P., & McArdle, G. (2015). Knowing and
- governing cities through urban indicators, city benchmarking and real-time dashboards. *Regional Studies, Regional Science,* 2(1), 6–28. https://doi.org/10.1080/21681376.2014.983149
- Kohl, S., & Brunner, J. O. (2020). Benchmarking the benchmarks comparing the accuracy of Data Envelopment Analysis models in constant returns to scale settings. *European Journal of Operational Research*, 285(3), 1042–1057. https://doi.org/10.1016/j.ejor.2020.02.031
- Kourtit, K., Nijkamp, P., & Suzuki, S. (2021). Comparative urban performance assessment of safe cities through data envelopment analysis. *Regional Science Policy & Practice*, 13(3), 591– 603. https://doi.org/10.1111/rsp3.12276
- Kourtzanidis, K., Angelakoglou, K., Apostolopoulos, V., Giourka, P., & Nikolopoulos, N. (2021). Assessing impact, performance and sustainability potential of smart city projects: Towards a case agnostic evaluation framework. *Sustainability*, *13*(13), Article 7395. https://doi.org/10.3390/su13137395
- Kraidi, A. A., Daneshvar, S., & Adesina, K. A. (2024). Weight-restricted approach on constant returns to scale DEA models: Efficiency of internet banking in Turkey. *Heliyon*, *10*(10). Article e31008. https://doi.org/10.1016/j.heliyon.2024.e31008
- Kramers, A., Höjer, M., Lövehagen, N., & Wangel, J. (2014). Smart sustainable cities – exploring ICT solutions for reduced energy use in cities. *Environmental Modelling & Software*, 56, 52–62. https://doi.org/10.1016/j.envsoft.2013.12.019
- Kushwah, V. S., Parashar, J., Dabas, P., Meena, L., & Sharma, V. (2024). Data science advancements in healthcare, education, and cities: An overview. In T. Murugan, J. W., & V. P. (Eds.), *Technologies for sustainable healthcare development* (pp. 17– 36). IGI Global Scientific Publishing.

https://doi.org/10.4018/979-8-3693-2901-6.ch002

Kutty, A. A., Kucukvar, M., Abdella, G. M., Bulak, M. E., & Onat, N. C. (2022). Sustainability performance of European smart cities: A novel DEA approach with double frontiers. *Sustainable Cities* and Society, 81, Article 103777.

https://doi.org/10.1016/j.scs.2022.103777

- Lacson, J. J., Lidasan, H. S., Spay Putri Ayuningtyas, V., Feliscuzo, L., Malongo, J. H., Lactuan, N. J., Bokingkito, P., & Velasco, L. C. (2023). Smart city assessment in developing economies: A scoping review. *Smart Cities*, 6(4), 1744–1764. https://doi.org/10.3390/smartcities6040081
- Lai, C. S., Jia, Y., Dong, Z., Wang, D., Tao, Y., Lai, Q. H., Wong, R. T. K., Zobaa, A. F., Wu, R., & Lai, L. L. (2020). A review of technical standards for smart cities. *Clean Technologies*, 2(3), 290–310. https://doi.org/10.3390/cleantechnol2030019
- Lee, C., & Lee, E. H. (2024). Evaluation of urban nightlife attractiveness for Millennials and Generation Z. *Cities*, 149, Article 104934. https://doi.org/10.1016/j.cities.2024.104934
- Lee, E. H., Shin, H., Cho, S.-H., Kho, S.-Y., & Kim, D.-K. (2019a). Evaluating the efficiency of transit-oriented development using network slacks-based data envelopment analysis. *Energies*, *12*(19), Article 3609. https://doi.org/10.3390/en12193609
- Lee, E. H., & Jeong, J. (2023). Assessing equity of vertical transport system installation in subway stations for mobility handicapped using data envelopment analysis. *Journal of Public Transportation*, 25, Article 100074. https://doi.org/10.1016/j.jpubtr.2023.100074
- Lee, E. H., Lee, H., Kho, S.-Y., & Kim, D.-K. (2019b). Evaluation of transfer efficiency between bus and subway based on data envelopment analysis using smart card data. *KSCE Journal of Civil Engineering*, 23(2), 788–799.

https://doi.org/10.1007/s12205-018-0218-0

- Liu, D., & Chen, Q. (2022). A novel three-way decision model with DEA method. *International Journal of Approximate Reasoning*, 148, 23–40. https://doi.org/10.1016/j.ijar.2022.05.003
- Liu, X., Payakkamas, P., Dijk, M., & de Kraker, J. (2023). GIS models for sustainable urban mobility planning: Current use, future needs and potentials. *Future Transportation*, 3(1), 384–402. https://doi.org/10.3390/futuretransp3010023
- Lombardi, P., Giordano, S., Farouh, H., & Yousef, W. (2012). Modelling the smart city performance. *Innovation: The European Journal of Social Science Research*, 25(2), 137–149. https://doi.org/10.1080/13511610.2012.660325
- Lytras, M. D., & Visvizi, A. (2018). Who uses smart city services and what to make of it: Toward interdisciplinary smart cities research. *Sustainability*, *10*(6), Article 1998. https://doi.org/10.3390/su10061998
- Mahajan, V., Mogha, S. K., & Pannala, R. K. P. K. (2024). Evaluation of efficiency and ranking of Indian hotels and restaurants: A bootstrap DEA approach. *Benchmarking: An International Journal*, 31(1), 186–198. https://doi.org/10.1108/BIJ-07-2021-0443
- Makki, A. A., & Alqahtani, A. Y. (2024). Analysis of the barriers to smart city development using DEMATEL. Urban Science, 8(1), Article 10. https://doi.org/10.3390/urbansci8010010
- Manoharan, G., Durai, S., Rajesh, G. A., Razak, A., Rao, C. B. S., & Ashtikar, S. P. (2023). Chapter five: An investigation into the effectiveness of smart city projects by identifying the framework for measuring performance. In V. Basetti, C. K. Shiva, M. R. Ungarala, & S. S. Rangarajan (Eds.), Artificial intelligence and machine learning in smart city planning (pp. 71–84). Elsevier. https://doi.org/10.1016/B978-0-323-99503-0.00004-1
- Mao, C., Wang, Z., Yue, A., Liu, H., & Peng, W. (2023). Evaluation of smart city construction efficiency based on multivariate data fusion: A perspective from China. *Ecological Indicators*, 154, Article 110882. https://doi.org/10.1016/j.ecolind.2023.110882
- Mardani, A., Zavadskas, E. K., Streimikiene, D., Jusoh, A., & Khoshnoudi, M. (2017). A comprehensive review of data envelopment analysis (DEA) approach in energy efficiency. *Renewable*

and Sustainable Energy Reviews, *70*, 1298–1322. https://doi.org/10.1016/j.rser.2016.12.030

Milošević, M. R., Milošević, D. M., Stanojević, A. D., Stević, D. M., & Simjanović, D. J. (2021). Fuzzy and interval AHP approaches in sustainable management for the architectural heritage in smart cities. *Mathematics*, 9(4), Article 304.

https://doi.org/10.3390/math9040304

- Moghaddas, Z., Yousefi, S., Mohammadi, M., & Tosarkani, B. M. (2023). A hybrid returns to scale-DEA model for sustainable efficiency evaluation of urban transportation systems. *International Journal of Systems Science: Operations & Logistics*, 10(1), Article 2221364. https://doi.org/10.1080/23302674.2023.2221364
- Mora, L., Bolici, R., & Deakin, M. (2017). The first two decades of smart-city research: A bibliometric analysis. *Journal of Urban Technology*, *24*(1), 3–27.

https://doi.org/10.1080/10630732.2017.1285123

- Mora, L., Deakin, M., & Reid, A. (2019). Strategic principles for smart city development: A multiple case study analysis of European best practices. *Technological Forecasting and Social Change*, 142, 70–97. https://doi.org/10.1016/j.techfore.2018.07.035
- Moradi, H., Lotfi, F. H., & Rostamy-Malkhalifeh, M. (2025). Inverse data envelopment analysis models for inputs/outputs estimation in two-stage processes. *Decision Making: Applications in Management and Engineering*, 8(1), 82–107. https://doi.org/10.31181/dmame8120251091
- Mosannenzadeh, F., Bisello, A., Diamantini, C., Stellin, G., & Vettorato, D. (2017). A case-based learning methodology to predict barriers to implementation of smart and sustainable urban energy projects. *Cities*, *60*, 28–36.

https://doi.org/10.1016/j.cities.2016.07.007

- Nasser, N., Khan, N., Karim, L., ElAttar, M., & Saleh, K. (2021). An efficient time-sensitive data scheduling approach for wireless sensor networks in smart cities. *Computer Communications*, 175, 112–122. https://doi.org/10.1016/j.comcom.2021.05.006
- Neirotti, P., De Marco, A., Cagliano, A. C., Mangano, G., & Scorrano, F. (2014). Current trends in smart city initiatives: Some stylised facts. *Cities*, *38*, 25–36. https://doi.org/10.1016/j.cities.2013.12.010
- Neves, F. T., de Castro Neto, M., & Aparicio, M. (2020). The impacts of open data initiatives on smart cities: A framework for evaluation and monitoring. *Cities*, *106*, Article 102860. https://doi.org/10.1016/j.cities.2020.102860
- Ninčević Pašalić, I., Ćukušić, M., & Jadrić, M. (2021). Smart city research advances in Southeast Europe. *International Journal* of Information Management, 58, Article 102127. https://doi.org/10.1016/j.ijinfomgt.2020.102127
- OECD. (2025). Science, technology and innovation scoreboard. https://www.oecd.org/en/data/datasets/science-technologyand-innovation-scoreboard.html
- Omrani, H., Fahimi, P., & Mahmoodi, A. (2020). A data envelopment analysis game theory approach for constructing composite indicator: An application to find out development degree of cities in West Azarbaijan province of Iran. *Socio-Economic Planning Sciences*, 69, Article 100675. https://doi.org/10.1016/j.seps.2018.12.002

Ozkaya, G., & Erdin, C. (2020). Evaluation of smart and sustainable cities through a hybrid MCDM approach based on ANP and TOPSIS technique. *Heliyon*, *6*(10), Article e05052.

https://doi.org/10.1016/j.heliyon.2020.e05052

- Patrão, C., Moura, P., & Almeida, A. T. de. (2020). Review of smart city assessment tools. *Smart Cities*, 3(4), 1117–1132. https://doi.org/10.3390/smartcities3040055
- Pelton, J. N., & Madry, S. (2024). Space systems, quantum computers, big data and sustainability: New tools for the United Na-

tions Sustainable Development Goals. In Artificial Intelligence for Space: AI4SPACE (pp. 53–104). CRC Press. https://doi.org/10.1201/9781003366386-3

- Raith, A., Ehrgott, M., Fauzi, F., Lin, K.-M., Macann, A., Rouse, P., & Simpson, J. (2022). Integrating data envelopment analysis into radiotherapy treatment planning for head and neck cancer patients. *European Journal of Operational Research*, 296(1), 289–303. https://doi.org/10.1016/j.ejor.2021.04.007
- Romão, J., Kourtit, K., Neuts, B., & Nijkamp, P. (2018). The smart city as a common place for tourists and residents: A structural analysis of the determinants of urban attractiveness. *Cities*, 78, 67–75. https://doi.org/10.1016/j.cities.2017.11.007

Ruhlandt, R. W. S. (2018). The governance of smart cities: A systematic literature review. *Cities*, *81*, 1–23. https://doi.org/10.1016/j.cities.2018.02.014

- Sarparast, M., Hosseinzadeh Lotfi, F., & Amirteimoori, A. (2022). Investigating the sustainability of return to scale classification in a two-stage network based on DEA models. *Discrete Dynamics in Nature and Society, 2022*(1), Article 8951103. https://doi.org/10.1155/2022/8951103
- Sharif, R. A., & Pokharel, S. (2022). Smart city dimensions and associated risks: Review of literature. Sustainable Cities and Society, 77, Article 103542. https://doi.org/10.1016/j.scs.2021.103542
- Sharifi, A. (2019). A critical review of selected smart city assessment tools and indicator sets. *Journal of Cleaner Production*, 233, 1269–1283. https://doi.org/10.1016/j.jclepro.2019.06.172
- Sharifi, A. (2020). A typology of smart city assessment tools and indicator sets. Sustainable Cities and Society, 53, Article 101936. https://doi.org/10.1016/j.scs.2019.101936
- Shen, X., Gu, Y., Zhao, X., & Xu, J. (2022). A data envelopment analysis evaluation study of urban crowd sourcing competitiveness based on evidence from 21 Chinese cities. *Frontiers* in *Psychology*, *13*, Article 861841.

https://doi.org/10.3389/fpsyg.2022.861841

- Simonofski, A., Vallé, T., Serral, E., & Wautelet, Y. (2021). Investigating context factors in citizen participation strategies: A comparative analysis of Swedish and Belgian smart cities. *International Journal of Information Management*, *56*, Article 102011. https://doi.org/10.1016/j.ijinfomgt.2019.09.007
- Singh, R., Kukreja, D., & Sharma, D. K. (2023). Blockchain-enabled access control to prevent cyber attacks in IoT: Systematic literature review. *Frontiers in Big Data*, *5*, Article 1081770. https://doi.org/10.3389/fdata.2022.1081770
- Stübinger, J., & Schneider, L. (2020). Understanding smart city a data-driven literature review. *Sustainability*, 12(20), Article 8460. https://doi.org/10.3390/su12208460
- Tan, S. Y., & Taeihagh, A. (2020). Smart city governance in developing countries: A systematic literature review. *Sustainability*, 12(3), Article 899. https://doi.org/10.3390/su12030899
- Toli, A. M., & Murtagh, N. (2020). The concept of sustainability in smart city definitions. *Frontiers in Built Environment*, 6, Article 77. https://doi.org/10.3389/fbuil.2020.00077
- Toloo, M., Mensah, E. K., & Salahi, M. (2022). Robust optimization and its duality in data envelopment analysis. *Omega*, 108, Article 102583. https://doi.org/10.1016/j.omega.2021.102583
- Toloo, M., & Tichý, T. (2015). Two alternative approaches for selecting performance measures in data envelopment analysis. *Measurement*, 65, 29–40.

https://doi.org/10.1016/j.measurement.2014.12.043

Tone, K., & Tsutsui, M. (2010). Dynamic DEA: A slacks-based measure approach. *Omega*, *38*(3), 145–156.

https://doi.org/10.1016/j.omega.2009.07.003

United Nations. (2019). World urbanization prospects 2018. https:// www.un.org/development/desa/pd/sites/www.un.org.development.desa.pd/files/files/documents/2020/Feb/un_2018_wup_ highlights.pdf

Van Puyenbroeck, T., Montalto, V., & Saisana, M. (2021). Benchmarking culture in Europe: A data envelopment analysis approach to identify city-specific strengths. *European Journal of Operational Research*, 288(2), 584–597.

https://doi.org/10.1016/j.ejor.2020.05.058

- Vanolo, A. (2014). Smartmentality: The smart city as disciplinary strategy. Urban Studies, 51(5), 883–898. https://doi.org/10.1177/0042098013494427
- Worthington, A., & Dollery, B. (2000). An empirical survey of frontier efficiency measurement techniques in local government. *Local Government Studies*, 26(2), 23–52. https://doi.org/10.1080/03003930008433988
- Xiong, B., Zhang, Q., Tao, X., & Goh, M. (2024). Benchmarking with quasiconcave production function under variable returns to scale: Exploration and empirical application. *Expert Systems with Applications, 243*, Article 122888.

https://doi.org/10.1016/j.eswa.2023.122888

- Ye, F., Chen, Y., Li, L., Li, Y., & Yin, Y. (2022). Multi-criteria decisionmaking models for smart city ranking: Evidence from the Pearl River Delta region, China. *Cities*, *128*, Article 103793. https://doi.org/10.1016/j.cities.2022.103793
- Yigitcanlar, T., Kamruzzaman, M., Buys, L., Ioppolo, G., Sabatini-Marques, J., da Costa, E. M., & Yun, J. J. (2018). Understanding 'smart cities': Intertwining development drivers with desired outcomes in a multidimensional framework. *Cities*, *81*, 145– 160. https://doi.org/10.1016/j.cities.2018.04.003
- Yigitcanlar, T., Kamruzzaman, M., Foth, M., Sabatini-Marques, J., da Costa, E., & Ioppolo, G. (2019). Can cities become smart without being sustainable? A systematic review of the literature.

Sustainable Cities and Society, 45, 348–365. https://doi.org/10.1016/j.scs.2018.11.033

- Yigitcanlar, T., Kankanamge, N., & Vella, K. (2022). How are smart city concepts and technologies perceived and utilized? A systematic geo-Twitter analysis of smart cities in Australia. In *Sustainable smart city transitions* (pp. 133–152). Routledge. https://doi.org/10.4324/9781003205722-7
- Zanella, A., Bui, N., Castellani, A., Vangelista, L., & Zorzi, M. (2014). Internet of Things for Smart Cities. *IEEE Internet of Things Journal*, 1(1), 22–32. https://doi.org/10.1109/JIOT.2014.2306328
- Zarrin, M., & Brunner, J. O. (2023). Analyzing the accuracy of variable returns to scale data envelopment analysis models. *European Journal of Operational Research*, *308*(3), 1286–1301. https://doi.org/https://doi.org/10.1016/j.ejor.2022.12.015
- Zhang, Y., Zhang, Y., Zhang, H., & Zhang, Y. (2022). Evaluation on new first-tier smart cities in China based on Entropy method and TOPSIS. *Ecological Indicators*, 145, Article 109616. https://doi.org/10.1016/j.ecolind.2022.109616
- Zhang, Y., Liu, F., Gu, Z., Chen, Z., Shi, Y., & Li, A. (2019). Research on smart city evaluation based on hierarchy of needs. *Procedia Computer Science*, *162*, 467–474.

https://doi.org/10.1016/j.procs.2019.12.012

- Zhao, H., Guo, S., & Zhao, H. (2018). Comprehensive performance assessment on various battery energy storage systems. *Ener*gies, 11(10), Article 2841. https://doi.org/10.3390/en11102841
- Zubir, M. Z., Noor, A. A., Mohd Rizal, A. M., Harith, A. A., Abas, M. I., Zakaria, Z., & Bakar, A. F. (2024). Approach in inputs & outputs selection of Data Envelopment Analysis (DEA) efficiency measurement in hospitals: A systematic review. *Plos One*, *19*(8), Article e0293694.

https://doi.org/10.1371/journal.pone.0293694